DIGITALES ARCHIV

ZBW – Leibniz-Informationszentrum Wirtschaft ZBW – Leibniz Information Centre for Economics

Srhoj, Stjepan; Škrinjarić, Bruno; Radas, Sonja et al.

Book

Closing the finance gap by nudging : impact assessment of public grants for women entrepreneurs = Smanjenje razlike u financiranju kroz poticaj : evaluacija učinaka bespovratnih sredstava za žene poduzetnice

Provided in Cooperation with:

Institute of Economics, Zagreb (EIZ)

Reference: Srhoj, Stjepan/Škrinjarić, Bruno et. al. (2019). Closing the finance gap by nudging : impact assessment of public grants for women entrepreneurs = Smanjenje razlike u financiranju kroz poticaj : evaluacija učinaka bespovratnih sredstava za žene poduzetnice. Zagreb, Croatia : The Institute of Economics, Zagreb.

This Version is available at: http://hdl.handle.net/11159/3115

Kontakt/Contact

ZBW – Leibniz-Informationszentrum Wirtschaft/Leibniz Information Centre for Economics Düsternbrooker Weg 120 24105 Kiel (Germany) E-Mail: *rights[at]zbw.eu* https://www.zbw.eu/

Standard-Nutzungsbedingungen:

Dieses Dokument darf zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden. Sie dürfen dieses Dokument nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, aufführen, vertreiben oder anderweitig nutzen. Sofern für das Dokument eine Open-Content-Lizenz verwendet wurde, so gelten abweichend von diesen Nutzungsbedingungen die in der Lizenz gewährten Nutzungsrechte.

https://savearchive.zbw.eu/termsofuse

Terms of use:

This document may be saved and copied for your personal and scholarly purposes. You are not to copy it for public or commercial purposes, to exhibit the document in public, to perform, distribute or otherwise use the document in public. If the document is made available under a Creative Commons Licence you may exercise further usage rights as specified in the licence.





Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics Stjepan Srhoj, Bruno Škrinjarić, Sonja Radas and Janette Walde

Closing the Finance Gap by Nudging: Impact Assessment of Public Grants for Women Entrepreneurs

Radni materijali ElZ-a - ElZ Working Papers

Svibanj . May 2019





Br. No EIZ-WP-1902

Radni materijali ElZ-a ElZ Working Papers ElZ-WP-1902

Closing the Finance Gap by Nudging: Impact Assessment of Public Grants for Women Entrepreneurs

Stjepan Srhoj

Teaching and Research Assistant Department of Economics and Business Economics, University of Dubrovnik Lapadska obala 7, 20 000 Dubrovnik, Croatia T. +385/ 20 445 936 E. ssrhoj@unidu.hr

Bruno Škrinjarić

Research Assistant The Institute of Economics, Zagreb Trg J. F. Kennedyja 7, 10 000 Zagreb, Croatia T. +385/ 1 2362 259 E. bskrinjaric@eizg.hr

Sonja Radas

Senior Research Fellow The Institute of Economics, Zagreb Trg J. F. Kennedyja 7, 10 000 Zagreb, Croatia T. +385/ 1 2362 200 E. sradas@eizg.hr

Janette Walde

Full Professor Department of Statistics, Faculty of Economics and Statistics, University of Innsbruck Universitaetsstr. 15, 6020 Innsbruck, Austria T. +43/ 512 507 704 22 E. janette.walde@uibk.ac.at

www.eizg.hr

Zagreb, May 2019

IZDAVAČ / PUBLISHER:

Ekonomski institut, Zagreb / The Institute of Economics, Zagreb Trg J. F. Kennedyja 7 10 000 Zagreb Hrvatska / Croatia T. +385 1 2362 200 F. +385 1 2335 165 E. eizagreb@eizg.hr www.eizg.hr

ZA IZDAVAČA / FOR THE PUBLISHER:

Maruška Vizek, ravnateljica / director

GLAVNA UREDNICA / EDITOR:

Ivana Rašić Bakarić

UREDNIŠTVO / EDITORIAL BOARD:

Ivan-Damir Anić Tajana Barbić Antonija Čuvalo Ivica Rubil Sunčana Slijepčević Paul Stubbs Marina Tkalec Maruška Vizek

IZVRŠNA UREDNICA / EXECUTIVE EDITOR: Doris Baničević

TEHNIČKI UREDNIK / TECHNICAL EDITOR: Vladimir Sukser

e-ISSN 1847-7844

Stavovi izraženi u radovima u ovoj seriji publikacija stavovi su autora i nužno ne odražavaju stavove Ekonomskog instituta, Zagreb. Radovi se objavljuju s ciljem poticanja rasprave i kritičkih komentara kojima će se unaprijediti buduće verzije rada. Autor(i) u potpunosti zadržavaju autorska prava nad člancima objavljenim u ovoj seriji publikacija.

Views expressed in this Series are those of the author(s) and do not necessarily represent those of the Institute of Economics, Zagreb. Working Papers describe research in progress by the author(s) and are published in order to induce discussion and critical comments. Copyrights retained by the author(s).

Sadržaj

	Abstract	5
1	Introduction	7
2	Theoretical Framework	10
3	Institutional Setting	12
4	Data and Methodology	13
4.1	Data	13
4.2	Method	14
4.3	Variables Used in the Analysis	16
5	Results	19
5.1	Descriptive Statistics and Matching	19
5.2	Average Treatment Effect on the Treated	20
5.3	Cost-Benefit Analysis	24
6	Discussion and Conclusion	25
7	Appendix	28
	Bibliography	36

Abstract:

Several recent papers draw attention to a lack of rigorous research on public policies supporting women entrepreneurs' competitiveness. This paper evaluates the effect of small business development gender-specific matching grants using a quasi-experimental approach. The grants have a positive effect on firm survival, as well as positive effects on obtaining bank loans, turnover, value added, employment, and total factor productivity. Heterogeneous treatment effects show that the grants increase the chance of young women entrepreneurs' firm survival and are even more effective for firms owned by mature women. Cost-benefit analysis estimates an increase in value added, which outweighs scheme-induced costs by 80% in the short-run and 170% in the long run.

Keywords: women entrepreneurship, public grants, policy evaluation, gender financing gap, behavioral additionality, nudging
JEL classification: B54, J16, H81, L26, L38, H43

Smanjenje razlike u financiranju kroz poticaj: Evaluacija učinaka bespovratnih sredstava za žene poduzetnice

Sažetak:

Nekoliko recentnih istraživanja skreće pozornost na nedostatak analiza utjecaja javnih politika koje potiču konkurentnost poduzeća u vlasništvu žena. Ovo istraživanje procjenjuje učinak javnog financiranja u obliku manjih bespovratnih potpora namijenjenih isključivo ženama poduzetnicama. Rezultati ukazuju na pozitivan utjecaj bespovratnih potpora na preživljavanje poduzeća, kao i pozitivne učinke na dobivanje bankovnih kredita te povećanje poslovnih prihoda, dodane vrijednosti, zaposlenosti i ukupne faktorske produktivnosti. Heterogeni učinci ukazuju na veću vjerojatnost preživljavanja poduzeća u vlasništvu mlađih žena, dok učinci na poduzeća u vlasništvu starijih žena poduzetnica ukazuju na ekonomski značajnije pozitivne efekte. Analiza troškova i koristi u kratkom roku procjenjuje 80% više stvorene dodane vrijednosti od troškova uzrokovanih potporom, dok se u dugom roku procjenjuje 170% više stvorene dodane vrijednosti od troškova uzrokovanih potporom.

Ključne riječi: žensko poduzetništvo, javne potpore, evaluacija politika, razlike između financiranja spolova; bihevioralna dodanost; poticaj
JEL klasifikacija: B54, J16, H81, L26, L38, H43

1 Introduction

Women make up about half of the world's population, but less than 30% of all businesses in Sweden or in the USA are women-owned, with percentages even lower in specialized, wealthy sectors such as technology (Ahl and Nelson, 2015). Nevertheless, women entrepreneurs are perceived as untapped potential for economic growth (Kauffman Foundation, 2011). Various studies in the entrepreneurship literature demonstrate the existence of gender differences in the way entrepreneurs approach and run their businesses (Foss et al., 2018; Henry et al., 2017; Jennings and Brush, 2013). Compared to men, women entrepreneurs are more sensitive to social, family-related and other non-financial concerns that significantly influence their business choices (Ahl and Nelson, 2015; Henry et al., 2017; Shahriar, 2018). Women tend to choose self-employment for the flexibility of working time, the workplace, or both (Allen and Curington, 2014). Thus, women entrepreneurs are overrepresented as owners of home-based part-time ventures (McAdam, 2012; Carter et al., 2015).

As opposed to their male counterparts, women entrepreneurs find it more difficult to acquire the necessary human, social and financial capital in order to establish and run their business (Henry et al., 2017). Women are less likely to be educated in science, technology, engineering or mathematics (STEM), which predestines them to work in low-technology sectors (Lerner et al., 1997; Marlow, 1997; Carter, 2000). Since entrepreneurs tend to set up their businesses in the same sector they had worked before, women entrepreneurs are more strongly represented in low-technology sectors, often characterized by relatively low growth. Not surprisingly, given this operating profile, women are also more likely to depend on informal sources of funding such as savings, credit and family support (Anyadike-Danes et al., 2015; Roberts, 2015; Lim and Suh, 2019).

Women are also less likely to hold executive positions prior to starting their own business, which deprives them of important leadership experience. Thus, compared to their male counterparts, they are less likely to belong to influential business networks that provide access to contacts, resources and relevant information. This lack of social capital (Neumeyer et al., 2018) also has additional negative consequences for accessing financial capital. Foss et al. (2018) show that not belonging to networks is a significant gender-related impediment for securing external financing.

Adequate financing is necessary for the survival, long-term growth and sustainability of small and medium-sized enterprises (SMEs), but women entrepreneurs face greater obstacles in securing financial capital than do male entrepreneurs (Marlow and Patton, 2005; Alsos et al., 2006; Orser et al., 2006; Verheul et al., 2012; Speelman et al., 2013; Roberts 2015; Cowling et al., 2019). Although obtaining capital involves both a supply side (lender) and a demand side (entrepreneur), most literature has focused on the former

inefficiencies (Wilson, 2009), while we focus on the latter and show how policy may be able to partly influence demand. It has been demonstrated that women show greater risk aversion to taking out a loan (Barber and Odean, 2001; Beckmann and Menkhoff, 2008; Hastings et al., 2013, Block et al., 2015), which is reflected in their lower demand for business finance (Croson and Gneezy, 2009; Huang and Kisgen, 2013). Interestingly, Cowling et al. (2019) argue that in the context of economic recession, prototypical forms of femininity actually become advantageous as banks seek to hedge risks by favoring lenders that are more conservative. Thus, we end up with the paradox that the risk aversion of women entrepreneurs prevents them from seeking financing, although it is precisely their risk aversion that may increase the chance of obtaining external financing. It would hence be beneficial to nudge women entrepreneurs to increase their demand for funding, for example by helping them to alleviate some of the excessive risk aversion.

In order to close the finance gap between men and women entrepreneurs, many governments have put in place gender-specific policies intending to help women-owned businesses (GEM, 2018). The policy dimension of women's entrepreneurship research has been neglected, as noted by Foss et al. (2018) in their systematic review of women's entrepreneurship policy. Examination of gender and entrepreneurship literature shows that only 4% of articles address public policy (Link and Strong, 2016). Consequently, very little is known about the impact of policies designed to support women entrepreneurs. Specifically, this pertains to overall policy effectiveness (Pandey and Amezcua, 2018), as well as to the impact on access to financial capital and/or the causal channels at work (Cho and Honorati, 2013; De Mel et al., 2014; Brush et al., 2018; Leitch et al., 2018).

This paper contributes to the literature concerned with supporting women entrepreneurship with policy measures. Using a quasi-experimental approach, we evaluate the impact of public grants with the aim to foster firms' development by affecting women entrepreneurs' demand for funding. We choose a policy instrument that provides very small amounts of money to the beneficiaries and argue that, despite the fact that these grants are too small to produce any direct effect on capital or labor, an observable impact can be accomplished by inducing a change in the beneficiary's behavior by nudging with the grant subsidized activities. The analyzed grant scheme is designed to give women entrepreneurs a nudge to more accurately evaluate their firm's situation and the risk involved in obtaining external financing. This more correct assessment of their firm situation may alleviate excessive risk aversion and encourage women entrepreneurs to seek appropriate funding. Thus, the policy achieves impact by changing behavior. This induced behavior additionality (Clarysse et al., 2009) should then have a measurable impact on the demand for funding and, consequently, on the firm's performance. To our knowledge, we are the first ones to investigate the impact on firms' competitiveness by nudging women entrepreneurs with small public grants towards more competition.

Specifically, we evaluate the grant scheme provided by the Ministry of Entrepreneurship and Crafts of the Republic of Croatia in the recession period from 2008 to 2012. Croatian figures for 2010 state that 37% of businesses were women-owned, but only 16% of women entrepreneurs were beneficiaries of public subsidies (European Commission, 2014; MINPO, 2014). The analyzed grants were given exclusively to women entrepreneurs and the focus was on micro and small enterprises. The financial level of support was too small (on average about 12,154 HRK1 or 1,639 EUR) to directly increase the capital or labor force of the firms. The grants supported activities such as preparing necessary documents for an investment, like a business plan, or getting childcare. More available time (resulting from not having to engage in childcare) or more expertise (thanks to the ability to purchase consulting services) enabled women entrepreneurs to better evaluate the economic situation of their firm and to plan necessary steps for firm development. This nudge may shift women entrepreneurs' willingness to enter into the competition for funding, for example by applying for bank loans. Implementation of entrepreneurial measures, e.g., getting a bank loan, can lead to a significant increase in a firm's output performance. Women entrepreneurs may also consider profits or other financing possibilities besides bank loans. This nudging process and the behavioral additionality as its result might be particularly relevant during the recession as the chance to obtain a bank loan might have been greater for women entrepreneurs (Cowling et al., 2019).

In order to empirically investigate the grant scheme we use a matching difference-indifferences estimation approach. We employ a rich dataset including financial and structural data on the universe of Croatian firms, data on the women entrepreneurship grants, demographic data on the entrepreneurs, and data on the employment history of entrepreneurs. The availability of the relevant covariates and the possibility for the matching approach to access all firms in Croatia should allow identification of the grant effects. Hence, based on the findings in the literature and from the causal channels of recent empirical and experimental studies, we hypothesize positive policy effects on firm survival, firm output, capital, employment, intermediate inputs, labor productivity, and total factor productivity (TFP). In addition, we examine whether grants have an indirect effect on capital through bank loans and whether women entrepreneurs' life experience implies heterogeneous grant impact effects.

The remainder of the paper is organized as follows: Section 2 presents the theoretical framework and Section 3 describes the institutional setting. Section 4 presents data and methodology followed by the results in Section 5. The last Section 6 provides the discussion and conclusions.

¹ 1 EUR ~ 7.41 Croatian kuna (HRK).

2 Theoretical Framework

For our theoretical framework, we use results from the empirical and experimental literature. The advantage of experiments conducted in the laboratory or in the field for the purpose of identifying causal effects is recognized in many substance sciences, especially recently also in the scientific literature in the entrepreneurship area (Williams et al., 2019). The causal findings of the relevant experimental studies as well as the economic theory of business decisions under uncertainty provide us with the theoretical framework for analyzing the effects of small public grants on the performance of women entrepreneurs in a quasi-experimental study.

Empirical evidence shows that women tend to be less competitive than men (Gneezy et al., 2003; Gneezy and Rustichini, 2004; Niederle and Vesterlund, 2007; Gupta et al., 2013; Niederle, 2015; Saccardo et al., 2017; Born et al., 2018). Experimental studies attempt to identify suitable framework conditions, under which women increase their willingness to compete. Gupta et al. (2013) show that women, when able to choose whom to compete with (men or women), tend to choose other women and then act more competitively in the experimental game. The authors explain this finding with an increased feeling of women's confidence. Recently, in an experimental study, Born et al. (2018) confirm that women in male-majority teams are less confident in their relative performance. They suggest that more targeted policies are needed to reduce this gap in competitiveness, such as affirmative action (Niederle et al., 2010), quotas or preferential treatment (Balafoutas and Sutter, 2010). The grants investigated in this research are in line with these results and are therefore awarded exclusively to women entrepreneurs.

Behavioural economic theories of decision-making suggest that women's competitiveness can be nudged towards more willingness to enter into competition. This is done by making a choice architectural change that alters people's behavior in a predictable way without forbidding any options or significantly changing their economic incentives. The analyzed policy intends to change women entrepreneurs' behaviour to actively engaging in competitive choices. One possible nudging approach is priming, referring to the temporal individual enabling of desired response behavior by making personal characteristics salient (Cohn et al., 2010; Cohn et al., 2014), activating particular feelings (Galinsky et al., 2003), and putting individuals in intended mind-sets (Gollwitzer et al., 1990). Applied in different contexts and on subject groups with a variety of socio-economic backgrounds, priming has been shown to successfully change how people think and behave (Frederiks et al., 2019; Rigtering et al., 2019). Balafoutas et al. (2018) show that priming with high power can significantly increase competition-entry decisions made by women. They argue that "when a feeling of power is activated, this might create in women a feeling of being in control to achieve one's goals and hence could close the gender gap in the willingness to compete" (p. 2). The analyzed grants are designed especially for women entrepreneurs in

order to increase the likelihood of a high participation rate. The grant was provided for activities to evaluate firms' development possibilities, with or without external expertise. After being more convinced of their firm's potential, women may be more willing to enter into competition. Investments, previously perceived as too risky, may subsequently be again in the choice set and even pursued further. The policy instrument analysed for the promotion of women entrepreneurs is thus designed in such a way that it leads to more competitive behaviour on the part of women and thus stimulates business growth.

Economic theory states that under uncertainty the optimal output for a risk-averse firm is characterized by marginal costs (C'(x)) being less than expected marginal revenue or price (E[p]) under full competition, i.e. $C'(x) \le E[p]$, where x is output and C(x) the variable cost function (Sandmo, 1971; Pope and Just, 1977; Blair 1984; Fooladi, 1985). Experiments demonstrate that women are more risk-averse than men and thus "gender" may impede investments profitable for the firm. Providing additional expertise or having more time available increases the relevant information set for the women entrepreneur, with which the expected value of the marginal revenue may increase. Thus, less risk-averse expectations encourage investments, and increases in the input and output performance of firms can be the result. The effect of a capital increase can be different depending on the firm's production function. If the women-owned firm is capital-constrained and has previously substituted labor for capital, it will reduce workers once it can buy capital to replace them. On the other hand, if capital and labor are complements, more capital will go along with hiring more labor.

In addition, we investigate heterogeneous effects with regard to young and mature women entrepreneurs. Prior studies show that the founder's previous professional experience has a significant influence on the performance of her firm (Colombo and Grilli, 2005), as individuals with higher industry-specific skills are better able to exploit business opportunities (Ganotakis, 2012). If this is the case, the grants should be more effective for mature women entrepreneurs as they are already aware of business opportunities.

In general, grants can affect firm output both directly and indirectly. Grants can directly ensure the inflow of needed capital that can boost firm output. They can also reduce information asymmetry, thus reducing the cost of capital and consequently increasing output via the certification effect (Martí and Quas, 2018; Hottenrott et al., 2018). In our case, the analyzed grant amounts are too small to produce any direct effects. The amount of the grant is also too small to have any effect via certification, and the grants are not provided for a concrete investment proposal, for which the bank is supposed to provide additional credit and, by awarding the grant, would receive the signal that it is worth actually granting the loan for this purpose. Therefore, any impact on company performance has to be achieved by nudging and thus by means of behavioral additionality.

3 Institutional Setting

The Republic of Croatia adopted the Strategy for Development of Women Entrepreneurship for the 2009–2013 and the 2014–2020 period, together with their Action Plans (MINPO, 2014). The aim of the grant strategy was to reduce the finance gap between women and men entrepreneurs. Table 1 gives evidence for the finance gap by providing the average amount of bank loans acquired by men versus women entrepreneurs. The Ministry of Entrepreneurship and Crafts of the Republic of Croatia (hereinafter: Ministry) was in charge of the vast majority of grant schemes in the Republic of Croatia². The grant schemes under the umbrella of this Ministry encompassed various competitiveness grant schemes (e.g. Srhoj et al., 2018). Women entrepreneurs did not make satisfactory use of the Ministry's public funding programs. In Croatia in 2010, 37% of all female entrepreneurs were women (European Commission, 2014), but only 19% benefited from public subsidies, although additional points were awarded to female entrepreneurs in the selection process (MINPO, 2014). In order to increase the allocation of grants to women, the Ministry launched several grant programs in the 2008–2012 period targeting exclusively women entrepreneurs of micro and small firms (details are given in Table A1 and Table A2 in the Appendix).

However, the effectiveness of these public funds has not yet been rigorously evaluated (MINPO, 2014). Critics argue that the government policies in Croatia were rather ineffective in spurring firm competitiveness (Bartlett, 2016; Srhoj et al., 2018). Bartlett (2016) concludes that policy measures did not have the desired impact on the competitiveness of Croatia's economy. He takes the Global Entrepreneurship Monitor's total early-stage entrepreneurial activity (TEA) index for women and men entrepreneurs to demonstrate considerable gender differences. The differences in the TEA index were rather large and persistent for the 2006–2014 period in Croatia (Bartlett, 2016), showing stagnation in the development of women entrepreneurship.

For the investigated women entrepreneurship grants potential recipients had to be registered in Croatia, had to have a surplus recorded in the previous year of business and at least one full-time employee, and had to have no unpaid debts towards the state or employees. The programs were matching grants, i.e. the grant co-financed 75% of the project value (VAT excluded), and firms had to finance the remaining 25% (and VAT) from private funds. Common activities that were co-funded by the grant included: (1) development of business plan and consulting services, (2) entrepreneurial training, (3) purchase of equipment, tools, and inventory, (4) preparing documentation for bank loans, and (5) babysitting and kindergarten costs. After the public call, firms submitted

² The policy basis for these grant schemes can be found in the policy documents entitled Operational Plan of Incentives for Small and Medium Entrepreneurship (OPPMSP, 2008; 2009; 2010; 2011) and Entrepreneurial Impulse: Plan of Incentives for Entrepreneurship and Craftsmanship (MINPO, 2012; 2013).

applications describing their planned activities along with their financial statements for the previous year. Evaluation was undertaken by the Ministry's expert team, taking into consideration the conditions elaborated above and the quality of the applicants. Table A2 in the Appendix shows that women entrepreneurship grant schemes encompassed 1,284 grants over the 2008–2012 period, with a total amount of 15,605,987 HRK (about 2,104,089 EUR) and on average 12,154 HRK (about 1,639 EUR) being awarded.

$\label{eq:table 1} \ \textbf{Differences in Amount of Bank Loans between Men- and Women-Owned Firms}$							
Industries (NACE 1-digit)	On	y men-owne	d firms	Only women-owned firms			
Industries (IACE I-digit)	N	Mean	S. D.	N	Mean	S. D.	
Agriculture, forestry, fishing	16,565	1,330,449	5,691,460	2,633	954,181	3,707,362	
Mining, quarrying	1,552	1,868,451	6,021,357	174	1,164,530	2,502,710	
Manufacturing	76,108	855,724	5,451,814	14,090	464,829	5,597,953	
Energy	2,321	3,552,665	24,636,720	275	1,355,619	9,884,572	
Water, sewage, waste	1,966	2,024,480	9,002,067	356	1,461,604	4,409,589	
Construction	80,010	2,332,674	53,477,844	10,049	953,948	6,367,973	
Wholesale, retail, motor vehicles	168,028	795,035	9,537,078	48,421	362,794	2,319,809	
Transport, storage	26,605	654,171	3,666,366	4,173	439,733	2,669,539	
Accommodation, food	31,865	1,564,572	23,992,582	8,727	376,811	1,935,207	
ICT	27,627	262,318	2,143,660	5,016	80,784	421,540	
Finance, insurance	3,961	9,815,577	95,349,906	1,051	153,148	1,400,915	
Real estate	14,614	4,570,686	26,275,716	4,425	615,516	7,030,754	
Professional, scientific work	75,336	815,551	18,401,673	35,458	97,966	877,045	
Administration	17,849	764,794	15,350,344	6,022	223,986	1,380,926	
Education	3,621	136,960	834,913	3,705	77,021	237,262	
Health, social work	3,892	418,914	1,802,891	3,252	225,530	1,081,743	
Arts, entertainment, recreation	5,115	1,084,389	8,595,110	1,256	859,086	13,543,615	
Other services activities	10,156	160,461	1,756,170	9,764	255,803	7,466,470	
TOTAL	567,191	1,211,396	24,636,601	158,847	347,138	3,797,945	

Notes: Data on the universe of Croatian firms, for which gender ownership structure data are available for the time period 2008 to 2011; N denotes number of firms, S.D. standard deviation.

4 Data and Methodology

4.1 Data

Five datasets were merged: financial and structural data on the population of Croatian enterprises for the 2005–2016 period, obtained from the Croatian Financial Agency (hereinafter: FINA); data on women entrepreneurship grants in the 2008–2012 period, obtained from the Ministry; data on demographic information on entrepreneurs, obtained from the court register of incorporated companies (hereinafter: Companies Register) and from the Ministry of Public Administration's birth registry (hereinafter: Birth Registry); and data on the employment history of entrepreneurs, obtained from the Croatian Pension Insurance Institute (hereinafter: HZMO).

The FINA dataset includes balance sheet and profit and loss statement data covering 300 variables for the universe of Croatian incorporated firms, as well as firm characteristics such as region, size, industry sector, firm ID, year of the report. On the other hand, the Ministry dataset includes the name of the grant recipient, the amount of grant given and the year the grant was received. The Companies Register contains identifiers for entrepreneurs, together with their characteristics such as age, gender, and their position within the firm. The HZMO dataset contains individual identifiers, firm identifiers, and the duration of their pension insurance within the firm. HZMO data are also connected to the Croatian Employment Office, which enabled us to obtain a person's total period of unemployment prior to receiving the grant. Finally, the Birth Registry dataset contains the identifiers for children associated with each person.

After merging the FINA and Ministry datasets, data are available on 196,300 firms, 1,274 of which obtained the analyzed grants. We remove all medium and large firms, foreignowned firms, and state-owned firms, as these were not eligible as recipients for the analyzed grants. For the same reasons, we remove all firms that reported a negative operating surplus, unpaid debts towards the state or employees, and firms with a male ownership structure in the year prior to treatment. We exclude firms having either no employees or zero turnover³. At this point, our sample is reduced to 59,299 firms. Firms receiving grants in the year of incorporation are dropped, as we have no previous financial records for these firms. Finally, we exclude all grant recipients in 2012 as these grants targeted a very small number of firms that were awarded heftier grant amounts than in previous grant schemes (Table A2 in the Appendix) and matching grants in 2012 had a 100% support as opposed to 75% support in the 2008–2011 period (Table A1 in the Appendix). Our analysis thus focuses on the 2008–2011 period only⁴. As identification strategy, we exclude firms also receiving other grants in the 2008–2012 period. We end up with 484 grant-awarded firms (treated firms) and 6,380 potential control firms (considering the time dimension, we have 20,392 observations).

4.2 Method

The methodical approach identifies the causal effect by comparing outcomes between a treatment group and a control group (Rosenbaum and Rubin, 1983). Treatment is usually modelled as a binary variable D, taking the value 1 for the treated firms and 0 for the control (non-treated, counterfactual) firms. The greatest challenge is to find a control firm

³ These observations are mostly crafts. In line with existing papers (e.g. Vitezić, Srhoj and Perić, 2018), we discard crafts as no financial data are available for these organizations. Crafts report their income on the basis of the Income Tax Law (OG 177/2004), while limited liability firms must keep accounting records at a detailed level according to the Accounting Act, Croatian and International Financial Reporting Standards, and International Accounting Standards.

⁴ We also performed the analysis with 2012 data, and our results remained robust. These results are available on request.

that is as similar as possible for each treated firm as firms may systematically differ in both observable and unobservable characteristics (Heckman et al., 1998). To mitigate this problem, Rubin (1977) introduced the conditional independence assumption (CIA), stating that potential outcomes are independent of treatment assignment (i.e. that exposure to treatment can be considered random), given a set of observable covariates X, which are not affected by the treatment, i.e. Y(D=0), $Y(D=1) \perp D \mid X$, where Y(D) denotes the potential outcome.

Empirically, this allows each treated firm to be matched with one (or several) control firms that are as similar as possible in their pre-treatment characteristics. We estimate this similarity of treated and control firm using a propensity score. This propensity score is defined as the conditional probability of receiving treatment given pre-treatment characteristics and is estimated using a standard probit model. We restrict the propensity scores to the common support area, thus considering only firms in the intersection of the range of the propensity scores for treated and control firms. Finally, the control firm for each treated firm is selected using the nearest neighbor method (for the baseline scenario) without replacement. Once the control firms are matched to the treated firms, the average

treatment effect on the treated firms (ATT) is estimated via $ATT = \frac{1}{N^T} \sum_{i \in T} (y_i^T - y_i^C)$,

where N^T is the number of treated firms and $y_i^T(y_i^C)$ the observed outcome of the treated (control) firm *i*.

To check the robustness of our ATT findings, we conduct a placebo test and perform matching with various matching algorithms. For the placebo test we discard the treated group, make the control group from our main specification a *placebo-treated* group and repeat our main specification matching procedure. If the effects are due to the grants, the placebo treatment should have no effect on firm performance. Additionally, we perform this procedure 10,000 times to empirically obtain the distribution of the ATT estimates. In this way, we do not have to rely on the calculated standard errors (Abadie and Imbens, 2008) or on the normality distribution of the ATT estimates. Moreover, we conduct a sensitivity analysis regarding the matching approach. We estimate nearest neighbor matching with two, three and four control firms per treated firm; nearest neighbor matching with two, three and four control firms per treated firm but with a caliper set at 10% of the standard deviation of the estimated propensity score; radius matching with the same caliper; kernel matching and local linear regressions using the Epanechnikov kernel.

In order to investigate the grants' mediation effect exerted on capital growth by bank loans growth, a causal mediation analysis is conducted (Imai et al., 2010). The causal mediation effect for each firm i is defined as

$$\delta_i(D) \equiv E[Y_i(D=1,M_i(D=1)) - Y_i(D,M_i(D=0))]$$

where D gives the treatment status, Y is the potential outcome variable and M is the mediator. The average causal mediation effect (ACME) is the expected difference in the potential outcome when the mediator takes the value that would occur under the treatment condition as opposed to the control condition, while the treatment status itself is held constant. For estimation, we compute two linear models with capital growth on the lefthand side. On the right-hand side are D, control variables, and the mediation variable (in our case bank loans growth). In this way, the direct effect (ADE) of the treatment as well as the indirect effect (ACME) are estimated.

In causal mediation analysis, an assumption additional to CIA is necessary, in particular the sequential ignorability assumption. This assumption implies that the mediator is ignorable given the observed treatment and pre-treatment confounders. The ignorability of the mediator implies that among those firms that share the same treatment status and pre-treatment characteristics, the mediator can be regarded as if it were randomized. This assumption cannot be directly tested. However, sensitivity analysis (Imai et al., 2010) is conducted to investigate how robust the results are. The correlation of the residuals (ρ) of the linear models is varied from -1 up to 1, ACME estimated and checked for the ρ value ACME that changes its sign. In our case large values of ρ imply that the conclusion is plausible given even fairly large departures from the ignorability of the mediator.

4.3 Variables Used in the Analysis

Our dataset includes a rich set of covariates (Table 2). All relevant variables for the public call schemes as well as covariates considered as important in the literature are used to calculate the propensity score.

As covariates, we select relevant firm characteristics that impact not only the selection into treatment, but the outcome as well. Some are obvious, such as firm age and size (measured in number of employees and real turnover); other covariates are proxies for firm characteristics, like average wage. Firms that pay higher average wages, have on average larger capital and cash reserves and thus are more likely to be financially stronger. Entrepreneurs of such firms may either be not interested in applying for the grants, or if they do apply, they may "make more out of it" than do weaker firms. The same holds for firms that have some knowledge-related capabilities indicated by ownership of fixed intangible assets, because these may facilitate more sophisticated production processes, which lead to higher outputs. We also include variables for financial constraints of firms, such as real values of short-term liabilities towards employees, short-term liabilities towards the state, liabilities towards banks and debt ratio, because firms with higher financial

constraints are found to be more vulnerable (Musso and Schiavo, 2008; Stucki, 2013). This vulnerability can induce firms to seek public aid and is also likely to shape how well the firm uses the grant. Debt ratio is included in quadratic form, as firms with more debt might be more motivated to apply for a grant, while those with too large a debt ratio may be rejected. The set of covariates also contains measures of productivity, such as real value added per employee and total factor productivity. These covariates are also assumed to affect firm performance. Firms that are exporters tend to be more productive (Costa et al., 2017) and to have specific entrepreneurial skills and human capital (Brambilla et al., 2012) that can affect both receiving a grant and the potential outcomes. Therefore, we use export-to-turnover ratio and import-to-turnover ratio.

Table 2 Covariates Used for Matching					
Variable	Description				
Treatment variable					
grant	1 if the firm received any grant scheme funding, 0 otherwise				
Firm characteristics					
age	Age of the firm				
age_sq	Squared age of the firm				
ownership	Ownership of the firm: 1 – State, 2 – Private, 3 – Mixed				
tech_intensity	Sectors of economy based on technological intensity ^a : 1 – Agriculture and mining, 2 – High-tech manufacturing, 3 – Mid high-tech manufacturing, 4 – Mid low-tech manufacturing, 5 – Low-tech manufacturing, 6 – Energy, 7 – Construction, 8 – Knowledge-intensive high-tech services, 9 – Knowledge-intensive other services, 10 – Less knowledge-intensive services				
nuts2_region	NUTS2 ^b region of the firm classified as: 1 – Continental Croatia, 2 – Adriatic Croatia				
firm_şize	Size of the firm: 1 – Micro (1–9 employees), 2 – Small (10–49 employees)				
trade_orientation	Trade orientation of the firm: 1 – Exporter only, 2 – Importer only, 3 – Exporter and importer, 4 – Domestic market only				
team_size	Number of people listed for each firm in the Companies Register: 1 – One, 2 – Two, 3 – Three or more				
gender_combination	Gender combinations connected to each firm in the Companies Register: 2 – Only women, 3 – Men and women				
Firm performance characte	eristics ^e				
labor	ln(1 + number of employees)				
average_wage	ln(1 + real average personnel costs)				
capital	ln(1 + real tangible fixed assets)				
fixed_intangible_assets_d	dummy for positive fixed intangible assets				
cash_reserves	ln(1 + real cash and cash equivalents)				
debt_ratio	Debt ratio (real fixed + current liabilities / real total assets)				
debt_ratio_sq	Squared debt ratio				
liabilities_banks	ln(1 + real total liabilities towards banks)				
turnover	ln(1 + real turnover)				
intermediate_costs	ln(1 + real intermediate (material + energy) costs)				
export-turnover	Export-to-turnover ratio				
import-turnover	Import-to-turnover ratio				
rva	ln(1 + real value added)				
tfp ^d	In(Total factor productivity)				
Entrepreneur (owner) char	racteristics				
mean_age	Mean age of firm owner(s)				
mean_age _sq	Squared mean age of firm owner(s)				
kids	Number of children				
work_experience	In(total work experience) (years)				

unemployment	In(unemployment) (years)
previous_experience	Dummy for previous work experience in current NACE2 sector
employers	Number of different employers
Year	
year	Dummy for each year in our sample

Notes: ^a More specific definitions of these technology sectors are available on request. ^b According to https://ec.europa.eu/eurostat/web/nuts/background. ^c All monetary variables were deflated using year- and sector-(NACE 2-digit) specific Eurostat deflators with base in 2010. Value added was deflated using value added deflator. Intermediate inputs (raw materials and energy) were deflated using intermediate input deflator. All other monetary variables were deflated using output deflators. ^d Total factor productivity was estimated using Wooldridge (2009) methodology based on the production function approach using value added as output, labor and capital as inputs, and intermediate inputs to control for unobservables. As technologies used in the production process differ across different industries, TFP was estimated separately for each NACE 2-digit industry.

We also include characteristics of the women entrepreneur, such as age and employment history, as this relates to aspects of human capital, which affects firm performance and survival (Colombo and Grilli, 2005). Square of age is also included as the grant-awarding agency may view entrepreneurs with too little experience as being too risky for a grant and a saturation effect in this relationship may also be present. We include the number of children as a limiting factor for the working hours, but also because childcare was one of the subsidized activities. As is common in the relevant literature, we also control for year, ownership, region, sector, size, trade and composition with respect to gender of top management team.

Outputs are categorized in six groups (Table 3): firm survival, output growth, labor inputs growth, capital inputs growth, intermediate inputs growth, and productivity growth.

Table 3 Outcome Variables Used in the Analysis					
Variable		Description			
Firm survival					
Active on the market	$survive_{t+q}$	Dummy if firm is still on the market in year $t + q$, $q \in \{1,, 5\}$			
Output growth					
In sales (turnover)	growth turnover _{$t-1,t+q$}	Real turnover growth from $t - 1$ to $t + q$, $q \in \{1,, 5\}$ ln (sales _{<i>t</i>+q}) - ln (sales _{<i>t</i>-1})			
In value added	growth value added $_{t-1,t+q}$	Real value added growth from $t - 1$ to $t + q$, $q \in \{1,, 5\}$ In (value added _{<i>t</i>+q}) – In (value added _{<i>t</i>-1})			
Labor inputs growth	1				
In employees	growth employees _{$t-1,t+q$}	Number of employees growth from $t - 1$ to $t + q$, $q \in \{1,, 5\}$ ln (employees _{t+q}) – ln (employees _{t-1})			
Capital inputs grow	th				
In capital	growth capital _{$t-1,t+q$}	Real capital growth from $t - 1$ to $t + q$, $q \in \{1,, 5\}$ ln (capital _{<i>t</i>+<i>q</i>}) – ln (capital _{<i>t</i>-1})			
In bank loans	growth liabilities $banks_{t-1,t+q}$	Real total liabilities towards banks growth from $t - 1$ to $t + q$, $q \in \{1,, 5\}$ In (liabilities _{<i>t</i>+<i>q</i>}) – In (liabilities _{<i>t</i>-1})			

Intermediate inputs	growth	
In intermediate input costs	growth intermediate $costs_{t-1,t+q}$	Real intermediate inputs growth from $t - 1$ to $t + q$, $q \in \{1,, 5\}$ In (intermediate costs _{<i>t</i>+q}) – In (intermediate costs _{<i>t</i>-1})
Productivity growth		
In total factor productivity	growth $\text{TFP}_{t-1,t+q}$	Real total factor productivity growth from $t - 1$ to $t + q$, $q \in \{1,, 5\}$ ln (TFP _{<i>t</i>+q}) - ln (TFP _{<i>t</i>-1})
In labor productivity (value added)	growth labor productivity $_{t-1,t+q}$	Real labor productivity (value added) growth from $t - 1$ to $t + q$, $q \in \{1,, 5\}$ In (labor productivity _{<i>t</i>+q}) – ln (labor productivity _{<i>t</i>-1})

5 Results

5.1 Descriptive Statistics and Matching

Treated firms are less frequently found in less knowledge-intensive service sectors (33%) and more frequently in low-technology manufacturing sectors (9%) than are the control firms before matching and on average have acquired larger bank loans. Treated firms more frequently have positive intangible assets (32%), domestic-only trade status (11%), but are less frequently just exporting firms (Table A3).

Treated firms are more frequently composed of several entrepreneurs. In particular, 31% of treated firms have three or more entrepreneurs as compared to 11% of the potential control firms. At the same time, the treated firms are less frequently single-owned (35%) as compared to all other women-owned firms (58%). A total of 59% of treated and 72% of control firms are owned by women only, while the remaining firms in both subsamples have a gender combination in ownership structure. Interestingly, although women entrepreneurs receiving and those not receiving grants do not significantly differ in the total number of years of work experience, the former are more likely to have several times changed their job prior to starting their own firm. They are also more likely to capitalize on their experience by starting their firms in the same NACE 2-digit industry in which they previously worked. Finally, on average there is no difference in the number of children per woman entrepreneur between treated and control firms.

We used a probit model with a dummy variable indicating whether or not the firm i received the grant in time t, $t \in \{2008, ..., 2011\}$, and with all firm performance variables, firm characteristics and entrepreneur characteristics as independent variables (Table A3). To avoid the problem of simultaneity, the covariates enter the calculations with a lag of one period. Estimation results are provided in Table A4. We do not interpret the specific findings obtained with the probit model. The purpose of the probit model is to forecast the propensity score and not to interpret the coefficient estimates or their statistical significance. The quality of the matched sample is our main objective here.

The propensity score is then used to find the control group composed of the nearest neighbors to the treated firms. Table A3 shows no significant differences in means of all covariates after matching and considerable decreases in standardized bias. The observed empirical densities of the covariates and the propensity score can be accepted as sufficiently equal for treated and control firms. The necessary balancing property is thus achieved.

5.2 Average Treatment Effect on the Treated

Women entrepreneurship grants yield a positive effect on firm survival as compared to control firms five years after treatment (Table 4).

Table 4 Treatment Effects of Women Entrepreneurship Grants on Firm Performance							
Autoomo variablas	АП (S. e.)						
	t + 1	<i>t</i> + 2	<i>t</i> + 3	t + 4	<i>t</i> + 5		
Firm survival		-					
Active on the market	-0.008	-0.008	0.010	0.017	0.025*		
	(0.009)	(0.011)	(0.015)	(0.017)	(0.018)		
Output growth							
In sales (turnover)	0.077**	0.097*	0.085*	0.058	-0.049		
	(0.046)	(0.063)	(0.064)	(0.070)	(0.071)		
In value added	0.088**	0.087*	0.100*	0.066	-0.011		
	(0.044)	(0.058)	(0.063)	(0.070)	(0.068)		
Labor inputs growth							
In employees	0.047**	0.051**	0.029	-0.008	-0.009		
	(0.022)	(0.026)	(0.031)	(0.034)	(0.037)		
Capital inputs growth							
In capital	0.431***	0.293*	0.329*	0.199	-0.161		
	(0.166)	(0.198)	(0.221)	(0.228)	(0.233)		
In bank loans	0.674**	0.832**	0.673*	0.769**	0.441		
	(0.357)	(0.398)	(0.424)	(0.451)	(0.449)		
Intermediate inputs growth	1						
In intermediate input costs	0.079*	0.095*	0.113*	0.120*	0.034		
	(0.054)	(0.070)	(0.077)	(0.080)	(0.082)		
Productivity growth							
In total factor productivity	0.039	0.066	0.097*	0.057	-0.006		
	(0.059)	(0.059)	(0.062)	(0.060)	(0.056)		
In labor productivity (value added)	0.045	0.056	0.098**	0.070	-0.002		
	(0.058)	(0.055)	(0.059)	(0.057)	(0.053)		

Notes: *p < 0.1; **p < 0.05; ***p < 0.01, one-sided p values. Standard errors (S.e.) are based on Abadie and Imbens (2008).

Grants induce a positive output additionality in sales and value added in the short run (t+1, t+2, and t+3 years after receiving the grant). However, at period t+4 and t+5 these effects vanish. The treatment has a positive significant effect on employment at time period t+1 and t+2, suggesting a positive effect on labor input additionality in the short run only.

With regard to capital inputs, grants induce positive capital input additionality in the short run, too. Treated firms have higher growth rates in bank loans up to four years after undergoing treatment. Intermediate inputs show positive growth effects in years t+1 to t+4. With regard to productivity, grants yield additionality in labor productivity and TFP only in year t+3.

The placebo test with and without the normality assumption demonstrates the robustness of our findings. All ATT estimates for the placebo-treated firms are not statistically significantly different from zero (Table A5). Figure A1 in the Appendix shows the empirical distribution of the estimated ATTs for 10,000 replications of the placebo test computed for each statistically significant ATT in Table 4. The red line marks the estimated ATT reported in Table 4, while green dashed lines represent the top and bottom 10% of the distribution. All estimated ATTs are in the far-right tail of their distribution, supporting the conclusion that there remains only a small probability that they occur by chance. These findings confirm that our ATTs (in Table 4) are attributable to the grants. Table A6 in the Appendix shows the results of the alternative matching approaches; all findings confirm the robustness of the main results.

Table 5 shows the indirect effect of grants on capital growth through bank loans. The average causal mediation effect (ACME) is the part of the ATT that is transmitted through bank loans growth, while the average direct effect (ADE) is the direct effect of treatment on capital growth. We find a strong grant indirect effect. This implies that the grant induced a positive effect on bank loans, which accounted for capital growth. When the ADE is statistically insignificant, this implies that the treated group does not statistically significantly differ in capital growth from the control group in the absence of the mediator. This points out the importance of grant indirect effects that are achieved via bank loans as a source for external financing. We find that the indirect effect of grants is significant in t+1, but in size it is smaller than the direct effect. This gives evidence that women entrepreneurs on average do not use bank loans only to finance firm investments.

Table 5 Mediation	Results of Treat	ment Through	Bank Loans Gro	wth on Capital	Growth
Effect	<i>t</i> + 1	<i>t</i> + 2	<i>t</i> + 3	<i>t</i> + 4	<i>t</i> + 5
ACME	0.056**	0.084**	0.092**	0.123**	0.086*
ADE	0.372**	0.187	0.276*	0.052	-0.277
Total effect	0.428***	0.271*	0.368**	0.175	-0.191

Notes: Standard errors are obtained via bootstrapping (10,000 simulations). The following lagged controls (t-1) are included: age, age squared, debt ratio, debt ratio squared, log capital, log profits, trade status dummies, firm size, and sector dummies. *p < 0.1; **p < 0.05; ***p < 0.01, one-sided p values. The coefficients for the controls are omitted for the sake of brevity and are available on request.

The sensitivity analysis indicates that the sign of the ACME would be maintained unless the sensitivity parameter ρ is greater than 0.18 (*t*+1), 0.2 (*t*+2), 0.21 (*t*+3), 0.26 (*t*+4) and 0.25 (*t*+5). Hence, the findings are of limited robustness to possible unobserved confounders.

Finally, we consider heterogeneous effects due to age and its associated characteristics of female entrepreneurs. Therefore, we split the sample into women below and above 40 years of age and re-estimate the ATT (Table 6). We find that the grant effects are indeed different for these two groups. The differences are particularly striking for growth in capital inputs (particularly in bank loans), where young women entrepreneurs exhibit stronger effects. Consequently, we investigate the differences between the two groups more closely. Younger women entrepreneurs, as compared to mature women, own on average younger (7.9 years versus 12.2 years) and smaller firms (5.3 employees versus 6.3 employees). They are thus more likely to have shorter and weaker credit histories than mature women entrepreneurs. In addition, younger women entrepreneurs own less capital – the average tangible fixed assets (proxy for capital) are significantly smaller for young women (112,645 EUR) versus mature women entrepreneurs (159,675 EUR). The same holds for liabilities to banks (61,853 EUR versus 79,375 EUR). Hence, any nudging towards competitiveness and investment is expected to increase capital and consequently output.

Although the grants are seen to have stronger effects on the survival of young women's firms, all other effects (output growth, intermediate inputs growth, labor input and long-term productivity) are much more pronounced for mature women entrepreneurs. This can be explained by the fact that these women have worked longer in the same industry; 60% of them have continued working in the same sector, which is a statistically significantly higher proportion than for the young women entrepreneurs ($\chi^2=8.587$, p=0.003). They might have been able to build larger industry-specific human capital, superior business networks and business skills (Ganotakis, 2012). Against this background of the women entrepreneurs, the nudging effect is most successful. For the sake of completeness, we investigate whether there is a difference between the two groups of women entrepreneurs regarding the number of children. In both groups more than 50% of the women entrepreneurs have no children (75th percentile is 1 child), and the proportion of women having children does not statistically differ between young (40%) and mature women entrepreneurs (45%, $\chi^2=1.8914$, p=0.169).

Table 6 Treatment Effects for Women he	low or 40 Ye	ars of Age ar	d above 40	Years of Ag						
		0		p	ATT (S	i. e.)				
Outcome variables	Wo	men below or 4	40 years of ag	$(n_T = 169^{a})$	-		Women a	ge > 40 (n_T =	= 315 ^a)	
	t + 1	t+2	t+3	t + 4	<i>t</i> + 5	t + 1	<i>t</i> + 2	<i>t</i> + 3	t + 4	<i>t</i> + 5
Firm survival										
Active on the market	0.006	0.012	0:030	0.041*	0.047*	-0.006	-0.003	0.013	0.013	0.022
Output growth										
In sales (turnover)	-0.003	0.014	0.010	0.008	-0.128	0.081**	0.076*	0.108**	0.170**	0.110*
In value added	-0.020	-0.060	-0.057	-0.031	-0.145	0.095**	0.105**	0.107**	0.162**	0.162**
Labor inputs growth										
In employees	0.051	0.045	0.019	0.003	-0.050	0.051**	0.075***	0.053*	0.064*	0.043
Capital inputs growth										
In capital	0.605**	0.525*	0.518*	0.598*	0.187	0.288*	0.272	0.167	-0.015	-0.046
In bank loans	1.777***	2.317***	1.387**	1.687**	1.148^{*}	0.497	0.723*	0.981**	0.964**	0.610
Intermediate inputs growth										
In intermediate input costs	0.040	-0.001	-0.012	0.072	-0.014	0.114**	0.132**	0.142**	0.194**	0.149**
Productivity growth										
In total factor productivity	-0.089	-0.079	0.027	-0.060	-0.151*	0.056*	0.037	0.055	0.100*	0.131**
In labor productivity (value added)	-0.094	-0.106	0.005	-0.034	-0.095	0.044	0.029	0.053	0.092*	0.119**

e do	
y w	
ina.	
fbr	
ke o	
e sai	
r th	
!; foi	
fied	
atis	
r is s	
bing	
atc	
r m	
afte	
erty	
rope	
id Si	
ncin	
ala	
). B	
008	
Ĩ	
ben	
Im	
and	
die i	
1 bac	
H uc	
ed a	s.
bas	îrm
are	ed f
e.)	reat
S (S.	oft
rror	ber
rd e	шті
ıdai	he n
Star	es ti
tes.	enoi
valı	$T_T d$
d_{p}	t. a 1
ideı	inesi
ne-s	req
<i>I</i> , o	е оп
0.0	labl
~ q	vai
***	re a
5.	ey a
0.0	t t b
> d	, bu
*	ults,
0.1;	sar :
v	t the
1 * .	por
otes.	t re
Ž	01

5.3 Cost-Benefit Analysis

The estimated treatment effects for women below or equal to and above 40 years of age (Table 6) enable us to make a cost-benefit analysis with common "*back-of-the-envelope*" calculation (Czarnitzki and Lopes-Bento, 2013). On the cost side, the amount of public funds provided for women entrepreneurs' business development projects was 0.2 million EUR and 0.4 million EUR for young and mature women entrepreneurs, respectively (2008–2011 period, Table A2(b)). The grant scheme had a positive effect on obtaining bank loans in the value of 3,008 EUR and 4,910 EUR for younger and mature women, respectively, per firm at t+2, as indicated in Table 7. This value multiplied by the number of total public grants awarded (169 and 315 for younger and mature women, respectively) gives a 0.5 million EUR and a 1.5 million EUR higher value of financing due to the grant scheme for younger and mature women, respectively. In sum, the grant scheme costs (0.2 million EUR and 0.4 million EUR) and the bank loans (0.5 million EUR and 1.5 million EUR) for the women entrepreneurship project amount to 0.7 million EUR and 1.9 million EUR for the younger and mature women, respectively.

On the yearly benefits side, the overall grants and bank loans lead to an increase in value added per firm at time t+2 of 14,532 EUR for mature women, which amounts to total (multiplied by 315 grants awarded to mature women) estimated benefits of 4.6 million EUR, and outweighs the public costs and higher bank loans by 2.4 times two years after the grants were distributed⁵. The benefits that are 2.4 times higher than costs are similar to the three times higher value added created by an export promotion policy in Denmark (Munch and Schaur, 2018). Since no additional value added was created among young women entrepreneurs, the benefits from mature women entrepreneurs can also be compared to the costs of both young and mature women entrepreneurs. A comparison of the 4.6 million EUR additional value added created by firms owned by mature women entrepreneurs and the 2.6 million EUR additional project costs shows that value added outweighs costs by 1.8 times. In the long run, at t+5, again no additionality is found for firms owned by young women entrepreneurs, and stronger positive effects on firms owned by mature women entrepreneurs are seen. The 22,452 EUR additional value added created per firm owned by mature women entrepreneurs (multiplied by 315 grants) leads to additional 7.1 million EUR value added and outweighs the additional project costs for young and mature women entrepreneurs (2.6 million EUR) by 2.7 times in the long run.

⁵ We do not report this analysis for the subsample of women younger than 40 years of age, as for them we do not find a significant effect on the value added growth figures.

able 7 Quantification of Treatment Effects for the Women Entrepreneurship Grant Scheme by Women's Age						
Outcome variables	<i>t</i> + 1	<i>t</i> +2	<i>t</i> + 3	<i>t</i> + 4	<i>t</i> + 5	
Women below o	or 40 years o	of age (n_T =	169 ^a)			
Real turnover	_ ^b	-	-	-	-	
Real value added	-	-	-	-	-	
Number of employees	-	-	-	-	-	
Real capital	6,622	5,745	5,674	6,545	-	
Real total liabilities towards banks	2,307	3,008	1,801	2,190	1,490	
Real intermediate inputs	-	-	-	-	-	
Women above 40 years of age (n_T = 315 ^a)						
Real turnover	12,564	11,791	16,772	26,349	17,053	
Real value added	13,170	14,532	14,811	22,490	22,452	
Number of employees	0.256	0.377	0.267	0.318	-	
Real capital	5,784	-	-	-	-	
Real total liabilities towards banks	-	4,910	6,657	6,545	-	
Real intermediate inputs	7,011	8,113	8,733	11,987	9,165	

Notes: We estimate the effects for the sample of treated firms in our analysis. All monetary variables are expressed in EUR. 1 EUR ~ 7.42 Croatian Kuna (HRK). ^a n_T denotes the number of treated firms. ^b We report only significant effects.

6 Discussion and Conclusion

Not only is the absolute number of female entrepreneurs still low, but there is also a considerable financing gap in the average amount of bank loans granted to male- and female-owned firms. Policy-makers are aware of this fact and of the economic potential that can evolve from supporting female entrepreneurs. Many governments have started to tackle this issue with public policies (GEM, 2018). While these initiatives are seen as positive, several scholars have called for further research on the effectiveness of these policies (Foss et al., 2018; Pandey and Amezcua, 2018). The aim of this paper is to provide evidence on the effectiveness of a public business development grant scheme as a possible instrument for promoting women entrepreneurs and reducing this financing gap.

From empirical and experimental studies we know that women entrepreneurs differ from their male counterparts in several aspects of their entrepreneurial orientation (Goktan and Gupta, 2015). Many experimental studies provide indications as to how a women-specific funding scheme should be designed. However, these proposals have not yet been implemented and tested for their suitability for use in supporting women entrepreneurs. When women have the opportunity to choose with whom they want to compete, they choose other women over men, and the more competitive a female co-participant is expected to be the more competitive they become themselves (Gupta et al., 2013). Therefore, grants awarded exclusively to women entrepreneurs are advantageous if the aim is to increase grant applications and women's competitiveness. We analyze the impact of a public support scheme aimed exclusively at women entrepreneurs. Specific to our case is the fact that the grant amount itself is too small to be invested directly in capital or labor as a means of increasing business performance. Rather, the grant serves as an incentive, as a so-called nudge, for female entrepreneurs to reassess the economic situation of their firm or possible investment opportunities. Activities such as the development of business plans, entrepreneurial training and obtaining childcare are supported. This allows women entrepreneurs either to spend more time in the company or to use external expertise to obtain information. When female entrepreneurs have been able to gather more information (and thus reduce their risk aversion), investments are more likely to be made. These investments can be observed in firm input and output performance measures. The main objective of this paper is to investigate whether the nudge provided by the design of the grant scheme was successful, i.e. behavioral additionality was stimulated and increases in firm performance were achieved.

The results show the grant scheme had a positive impact on firm survival. Positive effects on firm survival over the long run are in line with the findings of De Mel et al. (2013). However, in our case the positive effect on firm survival is driven by the impact that young women entrepreneurs have on firms.

For the entire sample, we find increased output additionality, but the effects do not persist on average beyond the third year after receiving a grant. Empirical research on long-term output additionality produced by subsidies is lacking, but the literature suggests that longterm effects may depend on the type of subsidy and characteristics of the entrepreneur. Goerke and Albers (2016) consider several types of subsidies and find long-term effects only for R&D instruments. De Mel et al. (2012) find long-term effects of small grants, too. These grants led to relatively large increases in business profits for male owners of microenterprises, but to no change in business profits for female business owners. They conclude "Capital alone thus does not appear to be enough to grow subsistence-level female-owned firms. Ongoing work is exploring the extent to which complementary interventions such as business training can help, or whether the other duties such as household production and child care constrain the extent to which women wish to grow their firms" (p. 965). The average age of the women entrepreneurs in the study by De Mel et al. (2012) is just around our cut-off point for investigation of heterogeneous effects (40 years). The long-term effect on firm output measures, however, is evident in our case for mature women entrepreneurs only. This can be explained by the higher industry-specific human capital of mature women entrepreneurs that allows them to better exploit business opportunities. We find statistical evidence for output additionality with respect to sales, value added and intermediate inputs growth. For younger women entrepreneurs, who have less capital, the behavioral additionality results in capital investments that do not pay off in the short term as compared to the control firms without grants, but that slightly increases the chance of firm survival.

The consideration of age and its associated characteristics of women entrepreneurs is decisive for impact assessment, as the heterogeneity of the effects is pronounced. A limitation of this study is that the sample size permitted us to split our sample only into young and mature women entrepreneurs. Further splits may be necessary to more appropriately capture the gradient in the acquired human or financial capital. We must leave that for future research. In addition, in our exploration of the effect that social environment exerts on female entrepreneurs we are limited by the available data. Neither having a child or not nor the number of children is responsible for the difference in the effect between young and mature women entrepreneurs. However, we were not able to obtain data on more personal details of the applicants such as their family income, or partner's employment status and income. Therefore, we cannot rule out differences in such variables between the investigated groups.

Nevertheless, we have a universe of women-owned firms and women entrepreneur data, on which we implemented various estimation approaches in order to be confident about the robustness of the results. The fact that the grants were awarded during the recession could have been particularly conducive to the impact effects achieved. Therefore, whether the conclusions also apply in times of economic growth remains to be further analyzed.

We conduct a cost-benefit analysis of the grant scheme with standard "back-of-theenvelope" calculation in the short and long term. In the short term, the yearly benefits outweigh the grant costs (defined by public money plus additional bank loans) by 2.4 times for mature women entrepreneurs' and by 1.8 times for both young and mature women entrepreneurs' grant costs. The positive effects on mature women entrepreneurs increase in the long run and benefits outweigh the young and mature women entrepreneur grant costs by 2.7 times.

To sum up, this study (1) provides robust evidence for the positive impact of small business development grants aimed exclusively at women entrepreneurs, (2) shows that such a grant scheme designed to nudge women entrepreneurs is particularly effective for mature women entrepreneurs, and (3) shows that the benefits of such a grant scheme clearly outweigh the costs.

Acknowledgements

We are grateful for the comments and suggestions made by the participants at the *2019 Diana International Research Conference* (held on June 2-5, 2019 at Babson College in Wellesley, MA, USA), and at the *2019 Innovation Management, Entrepreneurship and Sustainability Conference* (held on May 30-31, 2019 at University of Economics, Prague, Czech Republic).

Table A1 Women	Entrepreneurship Grant S	cheme(s) Description	
Grant scheme name	Subsidized activities	Minimum conditions	Min. and max. amounts
Women entrepreneurship 2008; 2009; 2010; 2011; 2012	Co-financing the business plan development, consulting services, documentation for bank loans, entrepreneurial training (apart from study costs), business registration costs, purchasing equipment, tools, inventory, and protective equipment, babysitting costs, kindergarten costs, child educator costs or extended school stay costs for the period of one year.	Firms in bankruptcy or liquidation procedure and those with unpaid debts towards the state or employees are not eligible. Surplus in previous year (turnover higher than costs). Headquarters in the Republic of Croatia. At least 1 full-time employee (including the owner). If the firm received the grant before – a statement of funds spent as planned in the previous year.	The support intensity is up to 75% of project costs (VAT excluded). Min. amount was set at 673 EUR from 2009 onwards (there was no min. amount in 2008) and max. amount at 10,768 EUR. In 2012 min. amount was set at 6,732 EUR and max. amount at 20,190 EUR with 100% support intensity (VAT excluded).

Note: All monetary variables are expressed in EUR. 1 EUR - 7.42 Croatian kuna (HRK). Source: Authors based on OPPMSP (2008; 2009; 2010; 2011) and MINPO (2012; 2013).

Table	A2 Wo	omen Entre	preneurshi	p Grant	Scheme	(s) De	scriptive S	Statistics					
					(a) Overa	all							
Vear			All grants					Final sample					
Tear	Grants	Amount	Mean (S. d.)	Min.	Max.	Grants	Amount	Mean (S. d.)	Min.	Max.			
2008	526	571,539	1,087 (776)	296	5,385	217	230,764	1,063 (704)	296	5,385			
2009	98	226,254	2,309 (1,476)	673	6,058	23	52,504	2,283 (1,311)	673	6,058			
2010	305	416,953	1,367 (1,245)	673	10,770	117	164,445	1,405 (1,359)	673	10,770			
2011	331	438,232	1,324 (1,076)	673	6,731	127	150,645	1,186 (794)	673	6,731			
2012	24	448,553	18,691 (2,986)	9,932	20,194	11	202,648	18,424 (3,917)	9,932	20,194			
Total	1,284	2,100,876	-	-	-	495	800,989	-	-	-			
(b) By Women Age (Final sample only)													
Year	Women age below or equal to 40							Women age above 40					
	Grants	Amount	Mean (S. d.)	Min.	Max.	Grants	Amount	Mean (S. d.)	Min.	Max.			
2008	89	93,315	1,048 (622)	296	4,043	128	137,682	1,076 (759)	377	5,391			
2009	8	19,272	2,409 (1,165)	674	4,043	15	33,288	2,219 (1,419)	674	5,526			
2010	35	39,825	1,138 (745)	674	4,043	82	124,798	1,522 (1,540)	674	10,782			
2011	37	43,935	1,187 (940)	674	5,391	90	106,873	1,187 (733)	674	4,717			
2012	4	71,765	17,941 (4,549)	11,119	20,216	7	131,102	18,729 (3,875)	9,943	20,216			
Total	173	268,113	-	-	-	322	533,743	-	-	-			

Note: All monetary variables are expressed in EUR. 1 EUR - 7.42 Croatian kuna (HRK). Source: Authors based on OPPMSP (2008; 2009; 2010; 2011) and MINPO (2012; 2013).

Table A3 Balance of Covariates Used in	Analysis (Common	Support Applies Aft	ter Matchin	ĝ)				
	B	efore matching		Af	ter matching			
	Control	Treated	i	Control	Treated	:		
Variable	Mean (S. d.)	Mean (S. d.)	St. blas	Mean (S. d.)	Mean (S. d.)	St. blas	% of reduction in standardized hias	
	(n * T = 18,734)	(n = 484)	(0/)	(<i>n</i> = 484)	(<i>n</i> = 484)	(n/)		
Firm characteristics								
Age of firm	10.05 (5.85)	9.52 (6.10)**	-8.9	9.45 (6.08)	9.52 (6.10)	1.3	85.2	
Age of firm (squared)	135.36 (115.30)	127.87 (116.06)	-6.5	126.11 (119.33)	127.87 (116.06)	1.5	76.5	
Ownership ^a								
State	۹,	٩	۹,	٩	٩	۹.	٩	
Private	(20.0) 0.00	0.99 (0.05)	4.6	0.99 (0.08)	0.99 (0.05)	7.1	-53.9 °	
Other	0.01 (0.07)	0.01 (0.05)	-4.6	0.01 (0.08)	0.01 (0.05)	-7.1	-53.9°	
Industry sectors ^a								
Agriculture, forestry, fishing, mining	0.01 (0.12)	0.01 (0.06)*	-10.5	0.01 (0.05)	0.01 (0.06)	2.2	79.3	
High-tech manufacturing	0.01 (0.08)	0.01 (0.10)	4.9	0.01 (0.12)	0.01 (0.10)	-4.6	6.2	
Mid high-tech manufacturing	0.01 (0.11)	0.01 (0.11)	0.2	0.01 (0.08)	0.01 (0.11)	5.6	-2,638.9 °	
Mid low-tech manufacturing	0.03 (0.18)	0.03 (0.18)	-0.2	0.03 (0.17)	0.03 (0.18)	1.2	-579.8°	
Low-tech manufacturing	0.05 (0.22)	0.09 (0.28)***	13.7	0.07 (0.26)	0.09 (0.28)	5.7	58.3	
Energy	۹,	۹.	۹,	٩	٩	۹.	۹.	
Construction	0.05 (0.22)	0.04 (0.20)	-5.5	0.04 (0.20)	0.04 (0.20)	-1.0	82.4	
High-tech KIS ^d	0.02 (0.15)	0.02 (0.16)	1.5	0.02 (0.16)	0.02 (0.16)	0.0	100.0	
Other KIS	0.30 (0.46)	0.45 (0.50)***	31.6	0.45 (0.50)	0.45 (0.50)	1.3	95.9	
LKIS ^d	0.50 (0.50)	0.33 (0.47)***	-35.2	0.36 (0.48)	0.33 (0.47)	-5.1	85.5	
Regions ^a								
Continental	0.65 (0.48)	0.72 (0.45)***	14.0	0.71 (0.46)	0.72 (0.45)	2.2	84.0	
Adriatic	0.35 (0.48)	0.28 (0.45)***	-14.0	0.29 (0.46)	0.28 (0.45)	-2.2	84.0	
Firm size ^a								
Micro	0.82 (0.39)	0.80 (0.40)	-5.3	0.80 (0.40)	0.80 (0.40)	-0.5	90.1	
Small	0.18 (0.39)	0.20 (0.40)	5.3	0.20 (0.40)	0.20 (0.40)	0.5	90.1	
International exposure ^a								
Exporter only	0.74 (0.44)	0.69 (0.46)**	-9.8	0.70 (0.46)	0.69 (0.46)	-0.5	95.3	
Importer only	0.08 (0.26)	0.08 (0.27)	1.1	0.08 (0.27)	0.08 (0.27)	-0.8	31.9	
Exporter and importer	0.11 (0.32)	0.12 (0.33)	2.2	0.13 (0.33)	0.12 (0.33)	-1.9	12.0	
Domestic market only	0.07 (0.26)	0.11 (0.31)***	11.9	0.10 (0.30)	0.11 (0.31)	3.6	69.6	
Team members ^a								
One team member	0.58 (0.49)	0.35 (0.48)***	-47.0	0.34 (0.47)	0.35 (0.48)	3.0	93.7	
Two team members	0.31 (0.46)	0.33 (0.47)	5.4	0.32 (0.47)	0.33 (0.47)	2.2	59.1	
								စ္ပ

						•		r
Three or more team members	0.11 (0.31)	0.31 (0.46)***	51.1	0.34 (0.47)	0.31 (0.46)	-6.3	87.8	
Gender ownership combination ^a								
Women only	0.72 (0.45)	0.59 (0.49)***	-26.4	0.59 (0.49)	0.59 (0.49)	1.3	95.0	
Men and women	0.28 (0.45)	0.41 (0.49)***	26.4	0.41 (0.49)	0.41 (0.49)	-1.3	95.0	
Firm performance characteristics								
In(number of employees)	1.62 (0.77)	1.71 (0.73)**	11.8	1.69 (0.82)	1.71 (0.73)	1.7	86.0	
In(real average wage)	10.66 (0.65)	10.61 (0.75)	-7.0	10.66 (0.77)	10.61 (0.75)	-7.2	-3.2°	
In(real capital)	10.96 (3.75)	11.12 (3.47)	4.4	11.02 (4.04)	11.12 (3.47)	2.8	35.5	
Fixed intangible assets dummy	0.23 (0.42)	0.32 (0.47)***	20.5	0.33 (0.47)	0.32 (0.47)	-3.7	81.8	
In(real cash reserves)	10.15 (2.68)	10.39 (2.26)*	9.5	10.4 (2.53)	10.39 (2.26)	-0.3	97.3	
Debt ratio	0.61 (0.30)	0.59 (0.25)	-7.1	0.59 (0.26)	0.59 (0.25)	0.3	95.3	
Debt ratio (squared)	0.46 (0.38)	0.41 (0.30)***	-14.4	0.41 (0.3)	0.41 (0.30)	-1.2	91.7	
In (real liabilities towards banks)	5.95 (6.28)	7.14 (6.24)***	19.0	7.40 (6.27)	7.14 (6.24)	-4.1	78.2	
In(real turnover)	13.85 (1.53)	13.88 (1.47)	1.9	13.89 (1.68)	13.88 (1.47)	-0.8	57.3	
In(real intermediate costs)	13.13 (1.91)	13.01 (1.93)	-6.6	13.05 (2.03)	13.01 (1.93)	-2.3	65.2	
Export-to-turnover ratio	0.05 (0.18)	0.05 (0.18)	0.2	0.05 (0.18)	0.05 (0.18)	-2.2	-908.5°	
Import-to-turnover ratio	0.06 (0.17)	0.05 (0.15)	-2.4	0.05 (0.14)	0.05 (0.15)	1.1	54.2	
In(real value added)	13.78 (1.54)	13.82 (1.47)	2.2	13.85 (1.66)	13.82 (1.47)	-2.1	4.4	
In(total factor productivity)	11.94 (1.29)	11.80 (1.20)**	-11.2	11.83 (1.31)	11.80 (1.20)	-1.9	83.2	
Entrepreneur (owner) characteristics								
Entrepreneur age	44.85 (7.90)	42.93 (7.37)***	-25.1	43.10 (7.35)	42.93 (7.37)	-2.2	91.1	
Entrepreneur age (squared)	2,074.05 (702.86)	1,897.61 (629.44)***	-26.4	1,911.93 (628.17)	1,897.61 (629.44)	-2.1	91.9	
Number of children	0.70 (0.99)	0.73 (0.95)	3.5	0.71 (1.07)	0.73 (0.95)	2.1	38.4	
Total work experience (years)	1.78 (0.98)	1.84 (0.91)	7.1	1.89 (0.88)	1.84 (0.91)	-4.5	35.8	
Total unemployment (years)	0.18 (0.44)	0.19 (0.43)	1.7	0.18 (0.44)	0.19 (0.43)	2.1	-22.7 °	
Previous employment in the same sector dummy	0.53 (0.5)	0.57 (0.50)*	8.2	0.55 (0.50)	0.57 (0.50)	3.7	54.2	
Number of different employers	1.41 (1.26)	1.55 (1.45)**	10.3	1.54 (1.21)	1.55 (1.45)	0.9	91.1	
Year								
2008	0.24 (0.43)	0.45 (0.50)***	44.3	0.45 (0.50)	0.45 (0.50)	-0.4	0.09	
2009	0.22 (0.41)	0.05 (0.21)***	-52.1	0.04 (0.20)	0.05 (0.21)	1.9	96.4	
2010	0.26 (0.44)	0.24 (0.43)	-5.2	0.25 (0.43)	0.24 (0.43)	-1.4	72.8	
2011	0.27 (0.45)	0.26 (0.44)	-2.6	0.26 (0.44)	0.26 (0.44)	0.5	81.9	
								Ľ

Notes: * p < 0.1; ** p < 0.05; $*** p < 0.01.^a$ Difference in categorical variables was tested using the χ^2 test. ^b Not at all or after cleaning not in the sample. ^c Denotes the cases where the standardized bias increases; however, in all these cases the initial difference in means between treated and non-treated units is insignificant. Percentage reduction in standardized bias is computed as $\left(1-\frac{|st.\ bias\ after\ matching|}{|st.\ bias\ before\ matching|}\right) \times 100^{-4}$ KIS and LKIS denote knowledge-intensive services and less knowledge-intensive services.

2

Table A4 Results of the Probit Model		
Variable ^a	Estimated coefficients	Standard error
Age of firm	-0.033**	(0.015)
Age of firm (squared)	0.002**	(0.001)
In(number of employees)	-0.024	(0.066)
In(real average wage)	-0.200***	(0.043)
In(real capital)	-0.011	(0.008)
Fixed intangible assets dummy	0.165***	(0.048)
In(real cash reserves)	0.010	(0.010)
Debt ratio	1.655***	(0.326)
Debt ratio (squared)	-1.467***	(0.262)
In(real liabilities towards banks)	0.010**	(0.004)
In(real turnover)	0.130	(0.080)
In(real intermediate costs)	-0.089***	(0.032)
Export-to-turnover ratio	-0.066	(0.136)
Import-to-turnover ratio	-0.133	(0.164)
In(real value added)	0.155*	(0.090)
In(total factor productivity)	-0.155***	(0.056)
Entrepreneur age	0.016	(0.025)
Entrepreneur age (squared)	-0.000	(0.000)
Number of children	0.028	(0.021)
Previous employment in the same sector dummy	0.038	(0.049)
In(total work experience (years))	0.080***	(0.029)
In(total unemployment (years))	0.002	(0.052)
Number of different employers	-0.007	(0.020)
N	20,39	92 ^b
McFadden pseudo R ²	0.14	14

Notes: * p < 0.1; *** p < 0.05; *** p < 0.01, two-sided p values, mean variance inflation factor (VIF) is 10.86 (as mentioned in the text, we do not interpret the significance of the coefficients, we simply use the estimated probability score). ^a In this model we also control for effects of years, ownership, region, sector, size, trade exposure, gender, and team. However, for the sake of brevity and presentation purposes, these results are available on request. ^b There is a difference between the number of observations in Table A4 (20,392) and the number of observations in Table A3 (18,734 + 484 = 19,218), because we remove some observations after probit estimations, as they do not have common support.

Table A5 Results of Placeb	o Test for th	e Baseline M	odel		
Outcome variables			ATT (S. e.)		
	<i>t</i> + 1	<i>t</i> + 2	<i>t</i> + 3	<i>t</i> + 4	<i>t</i> + 5
Firm survival					
Active on the market	0.004	0.012	-0.002	0.002	0.014
	(0.008)	(0.011)	(0.015)	(0.018)	(0.020)
Output growth					
In sales (turnover)	-0.029	-0.067	-0.019	0.001	0.039
	(0.049)	(0.064)	(0.071)	(0.078)	(0.075)
In value added	-0.052	-0.080	-0.038	-0.033	0.003
	(0.048)	(0.071)	(0.073)	(0.077)	(0.070)
Labor inputs growth					
In employees	0.010	0.007	0.026	0.076	0.080
	(0.023)	(0.028)	(0.034)	(0.066)	(0.059)
Capital inputs growth					
In capital	0.057	0.095	0.010	0.181	0.515
	(0.162)	(0.185)	(0.206)	(0.226)	(0.432)
In bank loans	0.047	-0.148	-0.553	-0.664	-0.473
	(0.351)	(0.399)	(0.499)	(0.543)	(0.446)
Intermediate inputs growth					
In intermediate input costs	-0.067	-0.107	-0.127	-0.112	-0.035
	(0.065)	(0.099)	(0.107)	(0.090)	(0.090)
Productivity growth					
In TFP	-0.048	-0.075	-0.055	-0.071	-0.051
	(0.057)	(0.066)	(0.073)	(0.074)	(0.064)
In labor productivity (value added)	-0.055	-0.079	-0.073	-0.079	-0.049
	(0.056)	(0.063)	(0.070)	(0.070)	(0.062)

Notes: p < 0.1; ** p < 0.05; *** p < 0.01, one-sided p values. Standard errors (s. e.) are based on Abadie and Imbens (2008). Balancing property after matching is satisfied. We do not report this for brevity, but the results are available on request.

Table A6 Result C	comparison between B	aseline Moc	lel and Othe	r Matching A	Ngorithms						
		NN (1)	NN (2)	NN (3)	NN (4)	NN (2)	NN (3)	NN (4)	Radius	Kernel	LLR
Variables		Baseline				Caliper	Caliper	Caliper	Caliper	Epan.	Epan.
		(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)	(6)	(10)
Firm survival											
	ln $t+1$	-0.008	-0.004	-0.005	-0.005	-0.004	-0.005	-0.005	-0.005	-0.003	-0.005
	In <i>t</i> + 2	-0.008	-0.001	-0.001	0.002	-0.001	-0.001	0.002	0.004	0.010	0.006
	In <i>t</i> + 3	0.010	0.010	0.011	0.010	0.010	0.011	0.010	0.012	0.019**	0.013
	In <i>t</i> + 4	0.017	0.018	0.021*	0.021*	0.018	0.022*	0.021*	0.021**	0.031***	0.023*
	In <i>t</i> + 5	0.025*	0.033**	0.034**	0.035**	0.033**	0.034**	0.035**	0.035***	0.046***	0.037**
Output growth											
	In sales at $t + 1$	0.077**	0.047	0.072**	0.075**	0.039	0.064**	0.069**	0.062**	0.093***	0.075*
	In sales at t + 2	•760.0	0.074*	0.087**	0.096**	0.064	0.077**	0.087**	0.061*	0.085**	0.073
	In sales at $t + 3$	0.085*	0.080*	0.127***	0.114***	0.069*	0.115***	0.104***	0.098***	0.127***	0.111^{*}
	In sales at $t + 4$	0.058	0.057	0.087**	0.077*	0.044	0.075*	0.065*	0.087**	0.111***	0.102*
	In sales at $t + 5$	-0.049	-0.019	0.004	0.007	-0.031	-0.009	-0.004	0.026	0.049	0.042
	In value added at $t + 1$	0.088**	0.056*	0.074**	0.07**	0.047	0.065**	0.062**	0.059**	0.086***	0.072*
	In value added at t + 2	0.087*	0.063*	0.085**	0.087**	0.053	0.074**	0.078**	0.055*	0.077**	0.067
	In value added at t + 3	0.100*	0.096**	0.130***	0.118***	0.084*	0.118***	0.108***	0.086**	0.111***	0.097*
	In value added at t + 4	0.066	0.068	0.082*	0.064	0.057	0.071*	0.054	0.080**	0.102***	0.092
	In value added at $t + 5$	-0.011	0.008	0.023	0.023	-0.005	0.010	0.012	0.049	0.070*	0.065
Labor inputs growth											
	In employees at $t + 1$	0.047**	0.053***	0.055***	0.056***	0.052***	0.054***	0.056***	0.051***	0.056***	0.051**
	In employees at t + 2	0.051**	0.061***	0.068***	0.073***	0.058***	0.064***	0.071***	0.066***	0.072***	0.069***
	In employees at t + 3	0.029	0.043*	0.054**	0.052**	0.040*	0.051**	0.050**	0.051***	0.058***	0.055**
	In employees at t + 4	-0.008	0.031	0.026	0.022	0.027	0.022	0.019	0.026	0.033*	0.033
	In employees at $t + 5$	-0.009	0.017	0.016	0.013	0.011	0.011	0.008	0.019	0.030	0.033
Capital inputs grow	th										
	In capital at $t + 1$	0.431***	0.381***	0.402***	0.404***	0.368***	0.393***	0.400***	0.388***	0.444***	0.409***
	In capital at $t + 2$	0.293*	0.258*	0.245*	0.251*	0.240*	0.229*	0.241*	0.261**	0.329**	0.282*
	In capital at t + 3	0.329*	0.235	0.245*	0.204	0.221	0.224	0.187	0.235*	0.309**	0.265
	In capital at $t + 4$	0.199	0.163	0.214	0.156	0.152	0.199	0.145	0.238*	0.324**	0.281

		-	-	-	-	-	-	-	-	-	-	-	-				-	-		-		-
0.252	0.821**	0.904**	0.603*	0.484	0.257		0.077*	0.062	0.083	0.108	0.066		0.017	0.037	0.068	0.079	0.038	0.016	0.018	0.056	0.088*	0.046
0.300**	0.664***	0.748***	0.422*	0.300	0.081		0.097***	0.083**	0.107**	0.130***	0.081*		0.027	0.043	0.082***	0.086***	0.046	0.027	0.026	0.072**	0.097***	0.056*
0.233*	0.889***	0.941***	0.635**	0.479*	0.307		0.038	0.025	0.047	0.067*	0.027		0.008	0:030	0.062**	0.078**	0.037	0.006	0.010	0.048*	0.085***	0.043
0.102	0.745***	0.773***	0.557*	0.567*	0.323		0.047	0.053	0.076*	0.057	0.012		0.010	0.050	0.078**	0.060*	0.014	0.011	0.037	0.067**	0.072*	0.032
0.100	0.787***	0.827***	0.623**	0.538*	0.286		0.034	0.045	0.073*	0.069	0.021		0.025	0.038	0.088**	0.053	0.003	0.027	0.024	0.082**	0.065*	0.019
-0.013	0.777***	0.845***	0.578*	0.525*	0.315		-0.009	0.005	0.010	0.013	-0.008		0.022	0.029	0.068*	0.014	-0.027	0.024	0.019	0.066*	0.027	-0.017
0.080	0.711***	0.734**	0.551*	0.606*	0.308		0.072*	0.076*	0.104**	0.088*	0.038		0.017	0.058*	0.086**	0.066*	0.022	0.018	0.044	0.075**	0.078**	0.038
0.084	0.752***	0.788***	0.615**	0.585*	0.264		0.058	0.067	0.100*	0.100*	0.044		0.032	0.045	0.097**	0.059	0.012	0.034	0.032	•*060.0	0.072*	0.026
-0.021	0.743**	0.807**	0.570*	0.539*	0.262		0.016	0.024	0.033	0.044	0.00		0:030	0.037	0.077*	0.022	-0.018	0.032	0.026	0.074*	0.034	-0.009
-0.161	0.674**	0.832**	0.673*	0.769**	0.441		0.079*	0.095*	0.113*	0.120*	0.034		0.039	0.066	0.097*	0.057	-0.006	0.045	0.056	0.098**	0.070	-0.002
In capital at $t + 5$	In bank loans at $t + 1$	In bank loans at $t + 2$	In bank loans at $t + 3$	In bank loans at $t + 4$	In bank loans at $t + 5$	Intermediate inputs growth	In intermediate input costs at $t + 1$	In intermediate input costs at $t + 2$	In intermediate input costs at $t + 3$	In intermediate input costs at $t + 4$	In intermediate input costs at $t + 5$	Productivity growth	In TFP at $t + 1$	In TFP at $t + 2$	In TFP at $t + 3$	In TFP at $t + 4$	In TFP at $t + 5$	In labor productivity (va) at $t + 1$	In labor productivity (va) at $t + 2$	In labor productivity (va) at $t + 3$	In labor productivity (va) at $t + 4$	In labor productivity (va) at $t + 5$

treated firm. Columns 5, 6, and 7 perform nearest neighbor matching	opensity score. Column 8 performs radius matching within the same	f 0.06.	mserve space. They are available on request.
m 1 presents our baseline results. Columns 2 to 4 use nearest neighbor matching with 2, 3, and 4 control firms for each treated	nd 4 control firms for each treated firm within a caliper defined as 10% of the standard deviation of the estimated propensi	mns 9 and 10 use kernel matching and local linear regression matching using the Epanechnikov kernel and bandwidth of 0.06.	< 0.05; *** $p < 0.01$, one-sided p values. Standard errors are based on Abadie and Imbens (2008), but are omitted to conserve
Notes: Colur,	using 2, 3, ú	caliper. Colu	b < 0.1; ** p



Bibliography

- 1. Abadie, A., and Imbens, G. W. (2008). On the failure of the bootstrap for matching estimators. *Econometrica*, 76(6), 1537–1557.
- 2. Ahl, H., and Nelson, T. (2015). How policy positions women entrepreneurs: A comparative analysis of state discourse in Sweden and the United States. *Journal of Business Venturing*, 30(2), 273–291.
- 3. Allen, W.D., and Curington, W.P. (2014). The self-employment of men and women: what are their motivations? *Journal of Labor Research*, *35*(2), 143–61.
- Alsos, G., Isaksen, E. J., and Ljunggren, E. (2006). New venture financing and subsequent business growth in men- and Women's entrepreneurship policy research: a 30-year review of the evidence women-led businesses. *Entrepreneurship Theory and Practice*, 30(5), 667–686.
- 5. Balafoutas, L., Fornwagner, H., and Sutter, M. (2018). Closing the gender gap in competitiveness through priming. *Nature communications*, 9(1), 4359.
- 6. Balafoutas, L., and Sutter, M. (2010). Gender, Competition and the Efficiency of Policy Interventions. IZA Discussion Paper No. 4955.
- 7. Bartlett, W. (2016). The failure of government policies to drive entrepreneurial performance in Croatia, In: C.C. Williams and A. Gurtoo (Eds.) *Routledge Handbook of Entrepreneurship in Developing Countries* (pp. 99–112). Abingdon: Routledge.
- 8. Born, A., Ranehill, E., and Sandberg, A. (2018). A Man's World? The Impact of a Male Dominated Environment on Female Leadership. *The Impact of a Male Dominated Environment on Female Leadership (July 3, 2018)*.
- 9. Brambilla, I., Lederman, D., and Porto, G. (2012). Exports, Export Destinations, and Skills. *American Economic Review*, *102*(7), 3406–3438.
- 10. Brush, C., Edelman, L. F., Manolova, T., and Welter, F. (2018). A gendered look at entrepreneurship ecosystems. *Small Business Economics*, 1–16.
- 11. Blair, R. D. (1974). Random input prices and the theory of the firm. *Economic Inquiry*, *12*(2), 214.
- Buttner, E. H., and Rosen, B. (1989). Funding new business ventures: Are decision makers biased against women entrepreneurs?. *Journal of Business Venturing*, 4(4), 249– 261.
- 13. Cho, Y., and Honorati, M. (2013). *Entrepreneurship programs in developing countries: A meta regression analysis*. The World Bank.

- Colombo, M. G., and Grilli, L. (2005). Founders' human capital and the growth of new technology-based firms: A competence-based view. *Research policy*, 34(6), 795– 816.
- 15. Cohn, A., Fehr, E. and Marechal, M. A. (2014). Business culture and dishonesty in the banking industry. *Nature* 516, 86–89.
- 16. Cohn, A., Maréchal, M. A. and Noll, T. (2010). Bad boys: How criminal identity salience affects rule violation. *Rev. Econ. Stud.* 82, 1289–1308.
- 17. Cohn, A., and Maréchal, M. A. (2016). Priming in economics. Current Opinion in Psychology, 12, 17–21.
- 18. Costa, S. Pappalardo, C., and Vicarelli, C. (2017). Internationalization choices and Italian firm performance during the crisis. *Small Business Economics*, 48(3), 753–769.
- 19. Cowling, M., Marlow, S., and Liu, W. (2019). Gender and bank lending after the global financial crisis: are women entrepreneurs safer bets?. *Small Business Economics*, 1–28.
- 20. Clarysse, B., Wright, M., and Mustar, P. (2009). Behavioural additionality of R&D subsidies: A learning perspective. *Research Policy*, 38(10), 1517–1533.
- 21. Czarnitzki, D., and Lopes-Bento, C. (2013). Value for money? New microeconometric evidence on public R&D grants in Flanders. *Research Policy*, 42(1), 76–89.
- 22. Czarnitzki, D., and Delanote, J. (2015). R&D policies for young SMEs: input and output effects. *Small Business Economics*, 45(3), 465-485.
- 23. Datta Gupta, N., Poulsen, A., and Villeval, M. C. (2013). Gender matching and competitiveness: Experimental evidence. *Economic Inquiry*, 51(1), 816–835.
- De Mel, S., McKenzie, D., and Woodruff, C. (2012). One-time transfers of cash or capital have long-lasting effects on microenterprises in Sri Lanka. *Science*, 335(6071), 962–966.
- 25. De Mel, S., McKenzie, D., and Woodruff, C. (2014). Business training and female enterprise start-up, growth, and dynamics: Experimental evidence from Sri Lanka. *Journal of Development Economics*, 106, 199–210.
- Decramer, S., and Vanormelingen, S. (2016). The effectiveness of investment subsidies: evidence from a regression discontinuity design, *Small Business Economics*, 47(4), 1007–1032.
- 27. Denicolai, S., Ramusino, E. C., and Sotti, F. (2015). The impact of intangibles on firm growth. *Technology Analysis and Strategic Management*, 27(2), 219–236.

- 28. Dvouletý, O., and Blažková, I. (2019). The Impact of Public Grants on Firm-Level Productivity: Findings from the Czech Food Industry. *Sustainability*, *11*(2), 552.
- Eddleston, K., Ladge, J. J., Mitteness, C., and Lakshmi, B. (2016). Do You See What I See? Signaling Effects of Gender and Firm Characteristics on Financing Entrepreneurial Ventures. *Entrepreneurship Theory and Practice*, 40(3), 489–514.
- 30. Fay, M., and Williams, L. (1993). Gender bias and the availability of business loans. *Journal of Business Venturing*, 8(4), 363–376.
- 31. Fooladi, I. (1985). A general theory of the competitive firm under undercertainty. *Atlantic Economic Journal*, *13*(3), 41–49.
- 32. Foss, L., Henry, C., Ahl, H., and Mikalsen, G. H. (2018). Women's entrepreneurship policy research: a 30-year review of the evidence. *Small Business Economics*, 1–21.
- 33. Frederiks, A.J., Englis, B.G., Ehrenhard, M.L., Gren, A.J., (2019). Entrepreneurial cognition and the quality of new venture ideas: an experimental approach to comparing future-oriented cognitive processes. *Journal of Business Venturing*, 34(2), 327–347.
- 34. Ganotakis, P. (2012). Founders' human capital and the performance of UK new technology based firms. *Small Business Economics*, 39(2), 495-515.
- 35. Galinsky, A. D., Gruenfeld, D. H. and Magee, J. C. (2003). From power to action. J. Pers. Soc. Psychol. 85, 453
- 36. GEM (2018). Global Entrepreneurship Monitor Global Report 2017/2018.
- 37. Gneezy, U., and Rustichini, A. (2004). Gender and competition at a young age. *American Economic Review*, 94(2), 377–381.
- Gneezy, U., Niederle, M., and Rustichini, A. (2003). Performance in competitive environments: Gender differences. *The Quarterly Journal of Economics*, 118(3), 1049– 1074.
- Goerke, B., and Albers, S. (2016). Long-term effects of subsidies on firm growth: introducing the concept of outcome additionality. ZBW - Deutsche Zentralbibliothek für Wirtschaftswissenschaften, Leibniz-Informationszentrum Wirtschaft, Kiel und Hamburg.
- Gollwitzer, P. M., Heckhausen, H. and Steller, B. (1990). Deliberative and implemental mind-sets: cognitive tuning toward congruous thoughts and information. J. Pers. Soc. Psychol. 59, 1119–1127.
- 41. Goktan, A. B., & Gupta, V. K. (2015). Sex, gender, and individual entrepreneurial orientation: evidence from four countries. *International Entrepreneurship and Management Journal*, 11(1), 95–112.

- 42. Grilli, L. and Murtinu, S. (2018). Selective subsidies, entrepreneurial founders' human capital, and access to R&D alliances. *Research Policy*, 47(10), 1945–1963.
- 43. Harris, R., and Moffat, J. (2013). Intangible assets, absorbing knowledge and its impact on firm performance: theory, measurement and policy implications. *Contemporary Social Science*, 8(3), 346–361.
- 44. Heckman, J. J., Ichimura, H., and Todd, P. (1998). Matching as an econometric evaluation estimator. *The Review of Economic Studies*, 65(2), 261–294.
- 45. Henry, C., Orser, B., Coleman, S., and Foss, L. (2017). Women's entrepreneurship policy: a 13 nation cross-country comparison. *International Journal of Gender and Entrepreneurship*, 9(3), 206–228.
- 46. Henry, C., Orser, B., Coleman, S., Foss, L., and Welter, F. (2018). Policy Support for Women Entrepreneurs' Access to Financial Capital: Evidence from Canada, Germany, Ireland, Norway, and the United States. *Journal of Small Business Management*, 1–27.
- 47. Hottenrott, H., Lins, E., and Lutz, E. (2018) Public subsidies and new ventures' use of bank loans. *Economics of Innovation and New Technology*, 27(8), 786–808.
- 48. Imai, K., Keele, L., Tingley, D., and Yamamoto, T. (2011). Unpacking the black box of causality: Learning about causal mechanisms from experimental and observational studies. *American Political Science Review*, *105*(4), 765–789.
- 49. Imai, K., Keele, L., and Tingley, D. (2010). A general approach to causal mediation analysis. *Psychological methods*, *15*(4), 309.
- 50. Jennings, J. E., and Brush, C. G. (2013). Research on women entrepreneurs: challenges to (and from) the broader entrepreneurship literature? *The Academy of Management Annals*, 7(1), 663–715.
- 51. Kauffman Foundation (2011). Overcoming the gender gap: women entrepreneurs as economic drivers. Ewing Marion Kauffman Foundation.
- 52. Leitch, C., Welter, F., and Henry, C. (2018). Women Entrepreneurs' Financing Revisited: Taking Stock and Looking Forward. *Venture Capital*, 20(2), 103–114.
- 53. Lerner, M., Brush, C., and Hisrich, R. (1997). Israeli women entrepreneurs: An examination of factors affecting performance. *Journal of Business Venturing*, 12(4), 315–339.
- 54. Link, A. N., and Strong, D. R. (2016). Gender and entrepreneurship: an annotated bibliography. *Foundations and Trends in Entrepreneurship*, *12*(4–5), 287–441.
- 55. Marlow, S., and Patton, D. (2005). All credit to men? Entrepreneurship, finance, and gender. *Entrepreneurship Theory and Practice*, 29(6), 717–735.

- 56. Martí, J., and Quas, A. (2018). A beacon in the night: government certification of SMEs towards banks. *Small Business Economics*, 50(2), 397–413.
- McKenzie, D. (2017). Identifying and spurring high-growth entrepreneurship: Experimental evidence from a business plan competition. *American Economic Review*, 107(8), 2278–2307.
- 58. Michalek, J., Ciaian, P., and Kancs, D. A. (2016). Investment crowding out: Firmlevel evidence from northern Germany. *Regional Studies*, 50(9), 1579–1594.
- 59. Ministry of Entrepreneurship and Crafts of the Republic of Croatia (2014). Strategija razvoja poduzetništva žena u Republici Hrvatskoj 2014 2020. Zagreb.
- 60. Musso, P., and Schiavo, S. (2008). The impact of financial constraints on firm survival and growth. *Journal of Evolutionary Economics*, 18(2), 135–149.
- 61. Neumeyer, X., Santos, S. C., Caetano, A., & Kalbfleisch, P. (2018). Entrepreneurship ecosystems and women entrepreneurs: A social capital and network approach. *Small Business Economics*, 1–15.
- 62. Niederle M. (2015) Gender, in *Handbook of Experimental Economics*, Vol 2, ed Kagel J and Roth AE.
- 63. Niederle, M., and Vesterlund, L. (2007). Do women shy away from competition? Do men compete too much?. *The Quarterly Journal of Economics*, *122*(3), 1067–1101.
- 64. Niederle, M., and Vesterlund, L. (2010). Explaining the gender gap in math test scores: The role of competition. *Journal of Economic Perspectives*, 24(2), 129–144.
- 65. Orser, B. J., Riding, A. L., and Manley, K. (2006). Women entrepreneurs and financial capital. *Entrepreneurship Theory and Practice*, 30(5), 643–665.
- 66. Pope, R. D., and Just, R. E. (1977). On the competitive firm under production uncertainty. *Australian Journal of Agricultural Economics*, 21(2), 111–118.
- 67. Rigtering, J. C., Weitzel, G. U., and Muehlfeld, K. K. (2019). Increasing quantity without compromising quality: How managerial framing affects intrapreneurship. *Journal of Business Venturing*, 34(2), 224–241.
- 68. Roberts, A. (2015). Gender, financial deepening and the production of embodied finance: Towards a critical feminist analysis. *Global Society*, *29*(1), 107–127.
- 69. Rosenbaum, P. R., and Rubin, D. B. (1983). The central role of the propensity score in observational studies for causal effects. *Biometrika*, 70(1), 41–55.
- 70. Rubin, D. B. (1977). Assignment to treatment group on the basis of a covariate. *Journal of Educational Statistics*, 2(1), 1–26.

- 71. Saccardo, S., Pietrasz, A., and Gneezy, U. (2017). On the size of the gender difference in competitiveness. *Management Science*, *64*(4), 1541–1554.
- 72. Sandmo, A. (1971). On the theory of the competitive firm under price uncertainty. *The American Economic Review*, *61*(1), 65–73.
- 73. Shahriar, A. Z. M. (2018). Gender differences in entrepreneurial propensity: Evidence from matrilineal and patriarchal societies. *Journal of Business Venturing*, 33(6), 762–779.
- 74. Shaw, E., Carter, S., and Brierton, J. (2001). Unequal Entrepreneurship: Why Female Entrepreneurship is an Uphill Business. London: Industrial Society.
- 75. Srhoj, S., Škrinjarić, B., and Radas, S. (2018). Bidding against the odds? The impact evaluation of grants for young micro and small firms during the recession. Radni materijali EIZ-a, (2), 5–37.
- 76. Stucki, T. (2013). Success of start-up firms: the role of financial constraints. *Industrial and Corporate Change*, 23(1), 25–64.
- 77. Sutter, M., and Glätzle-Rützler, D. (2014). Gender differences in the willingness to compete emerge early in life and persist. *Management Science*, *61*(10), 2339–2354.
- 78. Vitezić, V., Srhoj, S., and Perić, M. (2018). Investigating industry dynamics in a recessionary transition economy. *South East European Journal of Economics and Business*, 13(1), 43–67.
- Wallsten, S. J. (2000). The effects of government-industry R&D programs on private R&D: the case of the Small Business Innovation Research program. *The RAND Journal of Economics*, 82–100.
- 80. Williams, D. W., Wood, M. S., Mitchell, J. R., and Urbig, D. (2019). Applying experimental methods to advance entrepreneurship research: On the need for and publication of experiments. *Journal of Business Venturing*, *34*(2), 215–223.
- 81. Wooldridge, J. M. (2009). On estimating firm-level production functions using proxy variables to control for unobservables. *Economics Letters*, *104*(3), 112–114.

Popis objavljenih Radnih materijala EIZ-a / Previous issues in this series

2019

EIZ-WP-1901 Vedran Recher: *Tobacco smuggling in the Western Balkan region: Exploring habits, attitudes, and predictors of illegal tobacco demand*

2018

- EIZ-WP-1805 Dubravka Jurlina Alibegović, Željka Kordej-De Villa and Mislav Šagovac: *Smart City Indicators: Can They Improve Governance in Croatian Large Cities?*
- EIZ-WP-1804 Iva Tomić and Ivan Žilić: *Working for 200 euro? The effects of traineeship reform* on labor market outcomes in Croatia
- EIZ-WP-1803 Bruno Škrinjarić, Jelena Budak and Edo Rajh: *The Perceived Impact of Government Regulation in Reducing Online Privacy Concern*
- EIZ-WP-1802 Stjepan Srhoj, Bruno Škrinjarić and Sonja Radas: *Bidding against the odds? The impact evaluation of grants for young micro and small firms during the recession*
- EIZ-WP-1801 Dubravka Jurlina Alibegović: *Porezna autonomija gradova u Hrvatskoj u razdoblju 2002.–2016.*

2017

- EIZ-WP-1706 Katarina Bačić, Ivana Rašić Bakarić and Sunčana Slijepčević: *Sources of productivity differentials in manufacturing in post-transition urban South-East Europe*
- EIZ-WP-1705 Martin Larsson: *EU Emissions Trading: Policy-Induced Innovation, or Business as Usual? Findings from Company Case Studies in the Republic of Croatia*
- EIZ-WP-1704 Marina Tkalec and Ivan Žilić: *Does Proximity to Conflict Affect Tourism: Evidence from NATO Bombing*
- EIZ-WP-1703 Nebojša Stojčić and Zoran Aralica: *Choosing Right from Wrong: Industrial Policy* and (De)industrialization in Central and Eastern Europe
- EIZ-WP-1702 Bruno Škrinjarić, Jelena Budak and Mateo Žokalj: *The Effect of Personality Traits* on Online Privacy Concern
- EIZ-WP-1701 Alexander Ahammer and Ivan Zilic: *Do Financial Incentives Alter Physician Prescription Behavior? Evidence from Random Patient-GP Allocations*

2016

- ElZ-WP-1609 Edo Rajh, Jelena Budak, Jovo Ateljević, Ljupčo Davčev, Tamara Jovanov and Kosovka Ognjenović: *Entrepreneurial Intentions in Selected Southeast European Countries*
- EIZ-WP-1608 Ivan Žilić: *General versus Vocational Education: Lessons from a Quasi-Experiment in Croatia*
- EIZ-WP-1607 Valerija Botrić: *Public vs. private sector wage skill premia in recession: Croatian experience*
- EIZ-WP-1606 Edo Rajh, Jelena Budak and Mateo Žokalj: *Personal Values of Internet Users: A Cluster Analytic Approach*
- EIZ-WP-1605 Simon Stickelmann: *The Influence of the European Union Consumer Protection Policy on Croatian Consumers*
- EIZ-WP-1604 Bojan Basrak, Petra Posedel, Marina Tkalec and Maruška Vizek: *Searching high* and low: Extremal dependence of international sovereign bond markets
- EIZ-WP-1603 Valerija Botrić and Iva Tomić: *Self-employment of the young and the old: exploring effects of the crisis in Croatia*
- EIZ-WP-1602 Vedran Recher: *Tougher Than the Rest? Relationship between Unemployment and Crime in Croatia*
- EIZ-WP-1601 Iva Tomić: What drives youth unemployment in Europe?

2015

EIZ-WP-1505	Petra Palić, Petra Posedel Šimović and Maruška Vizek: The Determinants of
	Country's Risk Premium Volatility: Evidence from Panel VAR Model
EIZ-WP-1504	Sunčana Slijepčević, Jelena Budak and Edo Rajh: <i>Challenging Competition at Public Procurement Markets: Are SMEs Too Big to Fail? The Case of BiH and Croatia</i>

- EIZ-WP-1503 Ivan Žilić: Effect of forced displacement on health
- EIZ-WP-1502 Vedran Recher, Jelena Budak and Edo Rajh: *Eye in the Sky: Contextualizing Development with Online Privacy Concern in Western Balkan Countries*

EIZ-WP-1501 Petra Posedel Šimović, Marina Tkalec and Maruška Vizek: *Time-varying integration in European post-transition sovereign bond market*

2014

EIZ-WP-1403	Jelena Nikolic, Ivica Rubil and Iva Tomić: <i>Changes in Public and Private Sector Pay Structures in Two Emerging Market Economies during the Crisis</i>
EIZ-WP-1402	Jelena Budak, Edo Rajh and Ivan-Damir Anić: <i>Privacy Concern in Western Balkan Countries: Developing a Typology of Citizens</i>
	Jolona Budak and Eda Daily. The Dublic Producement Systems & Publicase Coster

EIZ-WP-1401 Jelena Budak and Edo Rajh: *The Public Procurement System: A Business Sector Perspective*

2013

- EIZ-WP-1302 Valerija Botrić: *Identifying Key Sectors in Croatian Economy Based on Input-Output Tables*
- EIZ-WP-1301 Ivica Rubil: Accounting for Regional Poverty Differences in Croatia: Exploring the Role of Disparities in Average Income and Inequality

2012

- EIZ-WP-1205 Hrvoje Mirošević: Analiza razvojnih dokumenata Republike Hrvatske
- EIZ-WP-1204 Iva Tomić: The Efficiency of the Matching Process: Exploring the Impact of Regional Employment Offices in Croatia
- EIZ-WP-1203 Jelena Budak, Ivan-Damir Anić and Edo Rajh: *Public Attitudes towards Surveillance and Privacy in Western Balkans: The Case of Serbia*
- EIZ-WP-1202 Valerija Botrić: Intra-industry Trade between the European Union and Western Balkans: A Close-up
- EIZ-WP-1201 Jelena Budak and Edo Rajh: *Corruption Survey in Croatia: Survey Confidentiality* and Trust in Institutions

2011

- EIZ-WP-1104 Jelena Budak and Edo Rajh: *Corruption as an Obstacle for Doing Business in the Western Balkans: A Business Sector Perspective*
- EIZ-WP-1103 Alfio Cerami and Paul Stubbs: *Post-communist Welfare Capitalisms: Bringing Institutions and Political Agency Back In*
- EIZ-WP-1102 Marina Tkalec: *The Dynamics of Deposit Euroization in European Post-transition Countries: Evidence from Threshold VAR*
- EIZ-WP-1101 Jelena Budak, Ivan-Damir Anić and Edo Rajh: *Public Attitudes Towards Surveillance and Privacy in Croatia*

2010

- EIZ-WP-1003 Marin Božić: *Pricing Options on Commodity Futures: The Role of Weather and Storage*
- EIZ-WP-1002 Dubravka Jurlina Alibegović and Sunčana Slijepčević: *Performance Measurement at the Sub-national Government Level in Croatia*
- EIZ-WP-1001 Petra Posedel and Maruška Vizek: *The Nonlinear House Price Adjustment Process in Developed and Transition Countries*

2009

- EIZ-WP-0902 Marin Božić and Brian W. Gould: *Has Price Responsiveness of U.S. Milk Supply Decreased?*
- EIZ-WP-0901 Sandra Švaljek, Maruška Vizek and Andrea Mervar: *Ciklički prilagođeni proračunski saldo: primjer Hrvatske*

2008

- EIZ-WP-0802 Janez Prašnikar, Tanja Rajkovič and Maja Vehovec: *Competencies Driving Innovative Performance of Slovenian and Croatian Manufacturing Firms*
- EIZ-WP-0801 Tanja Broz: *The Introduction of the Euro in Central and Eastern European Countries – Is It Economically Justifiable?*

2007

EIZ-WP-0705 Arjan Lejour, Andrea Mervar and Gerard Verweij: *The Economic Effects of Croatia's Accession to the EU*EIZ-WP-0704 Danijel Nestić: *Differing Characteristics or Differing Rewards: What is Behind the Gender Wage Gap in Croatia?*EIZ-WP-0703 Maruška Vizek and Tanja Broz: *Modelling Inflation in Croatia*EIZ-WP-0702 Sonja Radas and Mario Teisl: *An Open Mind Wants More: Opinion Strength and the Desire for Genetically Modified Food Labeling Policy*EIZ-WP-0701 Andrea Mervar and James E. Payne: *An Analysis of Foreign Tourism Demand for Croatian Destinations: Long-Run Elasticity Estimates*



